

# Working Paper Series 255

# Too Big to Fail Perception by Depositors: an empirical investigation Raquel de F. Oliveira, Rafael F. Schiozer and Lucas A. B. de C. Barros January, 2011

					50 00.038.166/0001-0
Working Paper Series	Brasília	n. 233	Jan.	2011	p. 1-71

ISSN 1518-3548 CGC 00.038.166/0001-05

# Working Paper Series

Edited by Research Department (Depep) - E-mail: workingpaper@bcb.gov.br

Editor: Benjamin Miranda Tabak – E-mail: benjamin.tabak@bcb.gov.br Editorial Assistant: Jane Sofia Moita – E-mail: jane.sofia@bcb.gov.br Head of Research Department: Adriana Soares Sales – E-mail: adriana.sales@bcb.gov.br

The Banco Central do Brasil Working Papers are all evaluated in double blind referee process.

Reproduction is permitted only if source is stated as follows: Working Paper n. 233.

Authorized by Carlos Hamilton Vasconcelos Araújo, Deputy Governor for Economic Policy.

### **General Control of Publications**

Banco Central do Brasil Secre/Surel/Cogiv SBS – Quadra 3 – Bloco B – Edifício-Sede – 1° andar Caixa Postal 8.670 70074-900 Brasília – DF – Brazil Phones: +55 (61) 3414-3710 and 3414-3565 Fax: +55 (61) 3414-3626 E-mail: editor@bcb.gov.br

The views expressed in this work are those of the authors and do not necessarily reflect those of the Banco Central or its members.

Although these Working Papers often represent preliminary work, citation of source is required when used or reproduced.

As opiniões expressas neste trabalho são exclusivamente do(s) autor(es) e não refletem, necessariamente, a visão do Banco Central do Brasil.

Ainda que este artigo represente trabalho preliminar, é requerida a citação da fonte, mesmo quando reproduzido parcialmente.

### **Consumer Complaints and Public Enquiries Center**

Banco Central do Brasil Secre/Surel/Diate SBS – Quadra 3 – Bloco B – Edifício-Sede – 2° subsolo 70074-900 Brasília – DF – Brazil Fax: +55 (61) 3414-2553 Internet: http://www.bcb.gov.br/?english

# Too Big to Fail Perception by Depositors: an empirical investigation<sup>1</sup>

Raquel de F. Oliveira<sup>2</sup> Rafael F. Schiozer<sup>3</sup> Lucas A. B. de C. Barros<sup>4</sup>

The Working Papers should not be reported as representing the views of the Banco Central do Brasil. The views expressed in the papers are those of the authors and not necessarily reflect those of the Banco Central do Brasil.

### Abstract

We examine a run from smaller to the largest banks in Brazil during the international financial turmoil in late 2008. Since Brazilian banks had no exposure to subprime securitized loans, the run is an opportunity to observe depositor reaction to a shock that is exogenous to the domestic banking system. Our empirical strategy allows us to disentangle the too big to fail benefit from the other features continuously related to size. Our unique database allows us to observe the behavior of uninsured deposits and of different types of holders of certificates of deposits, such as individuals, institutional investors and non-financial corporations. Taken together, our results are consistent with the idea that depositors ran from the smaller banks to the largest banks because they believed the largest banks were too big to fail. We also find that institutional investors had an important role in inducing the behavior of other depositors during the crisis: banks that had relatively more deposits of institutional investors ex-ante suffered more deposit outflows throughout the crisis.

**Key Words**: banking system, financial crisis, too big to fail **JEL Classification**: G21, G28, F33

<sup>&</sup>lt;sup>1</sup> We thank Franklin Allen, Heitor Almeida, Charles Calomiris, Murillo Campello, Giacomo Nocera, Michael Roberts, Ricardo Schechtman, Alexandre Tombini and participants in the 2010 Financial Management Meeting and in the 2010 Brazilian Finance Meeting, seminar participants at the Wharton School – University of Pennsylvania, University of Illinois at Urbana-Champaign and Central Bank of Brazil for helpful suggestions and comments. All remaining errors rest with the authors. Raquel Oliveira and Rafael Schiozer gratefully acknowledge financial support from FAPESP – Fundação de Amparo à Pesquisa do Estado de São Paulo.

<sup>&</sup>lt;sup>2</sup> Deorf, Banco Central do Brasil and Fundação Escola de Comércio Álvares Penteado

<sup>&</sup>lt;sup>3</sup> Deorf, Banco Central do Brasil and Fundação Getulio Vargas

<sup>&</sup>lt;sup>4</sup> Universidade Presbiteriana Mackenzie and Universidade de São Paulo

"In trying to stabilize the financial system, we have led creditors of large financial institutions to expect that the government will protect them from losses, which in turn means they need not monitor risk-taking by these firms."

Charles Plosse, President of the Federal Reserve Bank of Philadelphia at a policy forum sponsored by the Philadelphia Fed on Dec. 4, 2009<sup>1</sup>.

The crisis that started in the U.S. subprime sector in 2007 led to bailouts of large financial institutions in the US and Europe in late 2008 and early 2009. These bailouts brought the too big to fail policy back to the spotlight. Governments usually justify bailing out large financial institutions based on the harmful consequences of the failure of one large institution for the whole financial system. On the other hand, Finance researchers agree that a too big to fail policy is harmful to the long-run financial market stability because it distorts competition in the banking sector and weakens the incentives for creditors to monitor banks, which leads to moral hazard (Kaufman, 1990; Stern and Feldman, 2004; Gropp, Hakenes and Schnabel, 2010; Keister, 2010).

The empirical literature has examined the effect of bail-out policies on the risk-taking of banks (Boyd and Runkle, 1993, Schnabel, 2009), and the response of the capital markets to banks becoming too big to fail (O'Hara and Shaw, 1990; Brewer and Jagtiani, 2009). The empirical literature has not yet examined – to our knowledge – if the perception of a too big to fail policy affects depositor behavior. In other words: do depositors favor banks considered too big to fail regardless of their risk fundamentals?

This paper addresses this question by examining a run from smaller banks to the largest banks in Brazil during the international financial turmoil in late 2008. This flight to size is an opportunity to observe depositor reaction to a shock that is exogenous to the domestic banking system, since Brazilian banks had no exposure to subprime securitized loans. Specifically, we investigate if the depositor behavior may be credited to the perception of an implicit too big to fail policy. Our empirical strategy aims at ruling out the different reasons why depositors may have benefitted from putting their money in larger banks. Previous empirical studies on market discipline have found that

<sup>&</sup>lt;sup>1</sup> This quote can be seen in Sloan (2009).

depositors favor larger banks, but they did not differentiate between the too big to fail effect and other potential benefits of size (eg. Maechler and McDill, 2006; Imai, 2006).

Our unique bank level information, collected in close consultation with the financial supervisor, enables us to compare the behavior of uninsured depositors versus the total set of depositors. In addition, information on the type of holder of certificates of deposit allows us to identify differences in behavior of institutional investors, individual investors and non financial firms.

Taken together, our results are consistent with the idea that depositors ran from the smaller banks (hereafter, other banks) to the largest banks of Brazil (hereafter, big banks) during the most critical period of the international financial crisis because depositors believed the largest banks were too big to fail. The run was triggered by the bad news from the global banking industry in late 2008 and it was reverted after the turmoil passed. This observation strengthens the interpretation that the run was not based on fundamentals. We observe *no* systematic difference in the change in deposits for big banks in normal times, indicating that the competitive advantage of being too big to fail is most valuable during a crisis. The run is more pronounced for uninsured depositors, rather than for the total set of depositors, as was already expected. Our estimates are both statistically and economically significant. They imply that, after controlling for several sources of bank heterogeneity, including asset size and economic fundamentals, big banks increased uninsured deposits, on average, by approximately 48 percentage points more than other banks during the critical stage of the crisis. This expected additional increment amounts to approximately 37 percentage points for total deposits.

Our analysis of changes in certificates of deposit (CDs) by investor type provides interesting results as well. The amount of CDs held by institutional investors in big banks increased 30% in the last 6 months of 2008, while in the other banks there was a decrease of almost 35% in the CDs held by these investors. We also observe a similar movement for CDs of non financial firms: an increase of 55% in big banks and a decrease of 11% in other banks. When we control for bank fundamentals and the degree of dependence on institutional investors, we find that big banks enjoyed a change in the amount of CDs of institutional investors that was 57 percentage points larger than other banks. The effect for the CDs of non-financial firms is lower than for institutional

investors, but still important: a 40 percentage points increase in the amount of certificates of deposits in big banks relative to the other banks.

Our inferences are robust to different specifications of the set of big banks; that is, whether they include only the largest government-owned banks or the largest privately-owned banks.

Our primary contribution to the literature is to show that even an implicit too big to fail policy may be readily perceived by depositors. Our findings also contribute to the financial crisis empirical literature by showing that the presence of institutional investors played an important role in the run in Brazil. We find that each incremental percent point of assets being funded by CDs of institutional investors resulted in an outflow of 2.8% of both uninsured and total deposits during the crisis (we find no such effect for the other periods). Interestingly, the flow in CDs held by non-financial firms was also negatively influenced by the reliance on institutional investors. This result is in line with the evidence from mutual funds outflow found in Chen, Goldstein and Jiang (2010) and with some features of the model for bank runs of Goldstein and Pauzner (2005).

Finally, since the run was reverted after the turmoil and it did not quite reflect depositors' reassessment of bank fundamentals, our findings seem to support the view that a run may occur when depositors learn information from other banking systems, which may serve as a noisy signal on domestic bank-specific information. This result is consistent with some features of the bank runs model of Chen (1999) and Chen and Hassan (2008).

This research contributes to the financial literature for two main reasons: (1) it disentangles the continuous benefits of size from the discontinuous benefit of being perceived as too big to fail; (2) it is, to the best of our knowledge, the first study to document a massive flow of deposits to large banks, that is best explained by depositors anticipating possible bailouts of the large banks rather than by worse fundamentals of the smaller banks.

The remainder of the paper proceeds as follows. Section I lays out the motivation for our study. Section II introduces the empirical strategy. Section III discusses the data and sample selection. Section IV presents the results, Section V describes some robustness checks and Section VI concludes.

# I. The Effects of the Global Financial Crisis on Brazil

We begin our analysis by illustrating how the Brazilian financial system was affected by the global financial crisis<sup>2</sup>. The global financial crisis reached a turning point in September 2008, when Lehman's bankruptcy induced losses to several counterparties and forced markets to reassess risks that were previously overlooked. Investors withdrew from the markets and liquidity dried up, affecting the real economy. Allen and Carletti (2010) argue that the most disruptive consequence of Lehman's bankruptcy was the signal it sent to the international markets that credit risk in the banking sector and financial industry was a serious concern. Most banks around the world lost substantial equity value and governments had to infuse capital to prevent failure (Beltratti and Stulz, 2009).

In October 2008, the US and European Governments strengthened deposit guarantee schemes to avoid bank runs. Ireland, Greece, Germany, Denmark, France and others offered blanket guarantees (Willman, 2008; Hall, 2008). In the US, Congress raised the deposit insurance from \$100,000 to \$250,000 per depositor, per institution, although well-informed depositors know they can have full coverage by putting their money in one institution under separately titled accounts. The UK raised deposit protection from £35,000 to £50,000 (Cumbo, 2008), which was not enough to stop a flight to safety by bank depositors. National Savings & Investments, a state-owned bank that offers 100 per cent government guarantee received record deposit inflows in the last quarter of 2008. (Warwick-Ching, 2009). HSBC, which was perceived as safe, had also record inflows into its UK deposit accounts (Ross, 2008).

The turbulence in the US and Europe in the last quarter of 2008 affected the distribution of deposits across banks in Brazil, although none of the domestic banks (large or small) had exposure to US securitizations of subprime loans. The largest banks experienced a significant inflow of deposits from the smaller banks. Figure 1 provides a snapshot of the flow of funds from the smaller banks into the largest banks of Brazil (big banks) by showing daily data of the market share of certificates of deposit (CDs). From mid-September to December 2008 the big bank's market share of certificates of deposit spiked almost 7 percentage points, reaching 82.5%. It remained in that level until mid-March 2009, when it began to go back to pre-crisis market share.

 $<sup>^{2}</sup>$  An overview of the events of the crisis can be seen in Brunnermeier (2009) and Taylor (2009).

# [Insert Figure 1 here]

In an attempt to provide liquidity to the smaller banks, the Brazilian Government took several measures, such as the reduction of reserve requirements and the design of a new certificate of deposit with a special guarantee of BRL 20,000,000 (equivalent to around nine million US dollars at the time) in March 2009<sup>3</sup>.

Another way to see the run from smaller banks to the big banks is by looking at the market share of the big banks of insured and uninsured deposits. Figure 2 presents semiannual information on the share of total, uninsured and insured deposits of the set of big banks. The results are in line with those presented in Figure 1. There was a significant rise of 5 percentage points in the otherwise relatively stable share of total deposits of the big banks, during the exacerbation of the crisis. Most of this rise is due to a spike of 15 percentage points in the market share of uninsured deposits.

# [Insert Figure 2 here]

The exacerbation of the global financial crisis in late 2008 also affected the real economy of Brazil through basically two major channels: (i) an increase in risk aversion tightened external financing conditions and (ii) a significant decline in international investment, followed by lower demand for regional exports and a drop in commodity prices (IMF, 2009a, b).

The constraint in external financing led to a significant devaluation of the Brazilian currency, the Real, in the last quarter of 2008. Despite the devaluation, there was not a process of currency substitution in Brazil: the deposit base grew 24.7% during the third and fourth quarter of 2008, as can be seen in Figure 3.

# [Insert Figure 3 here]

# II. Testing the Perception of Too Big To Fail

As mentioned before, the flight to the big banks in Brazil during the most critical stage of the global financial crisis is an opportunity to observe depositor reaction to a shock that is exogenous to the domestic banking system, since domestic banks had no subprime securitized loans and insignificant exposure to foreign debt.

<sup>&</sup>lt;sup>3</sup> The measures to reduce reserve requirements were taken along the following dates in 2008: September 24, October 2, 8, 13, 14, 15, 24 and 31, November 13 and 25 and December 19. Measures to change the discount window were taken in October 6, 9, 10 and 16. The creation of a new debt instrument with a special guarantee of BRL 20,000,000 (equivalent to around nine million dollars at the time) was taken in March 26 and 31/2009.

There are two important questions to be asked: (1) why was there a reaction of depositors in Brazil to the international financial turmoil? and (2) why was the reaction a flight from the smaller banks to the big banks? Though the answer to the first question is not formally addressed in this paper, we conjecture that the bad news from the global banking industry in late 2008 made depositors reassess risks in the banking sector. In many aspects, this conjecture is closely related to the models of bank runs presented by Chen (1999) and Chen and Hassan (2008).

There are basically two lines of argument to explain the origins of bank runs. One is based on the classical work of Diamond and Dybvig (1983) where bank runs are self fulfilling prophecies. Given the assumption of first-come, first-served, and costly liquidation of some long-term assets, one possible equilibrium is that depositors should rationally withdraw their funds if they believe that other depositors will withdraw. Another equilibrium occurs when no one believes a banking panic is about to occur, so depositors only withdraw their funds according to their liquidity needs. Which of these two equilibria occurs depends on random shocks or "sunspots".

The second line of argument to explain the origins of banking panics is based on the business-cycle. This view asserts that crises are not random events, or the result of "sunspots", but a natural consequence of the business cycle. If depositors receive information about an upcoming downturn in the cycle, they try to withdraw their funds because they expect a reduction in the value of bank assets and a rise in the probability of bank failures. There are several theoretical models consistent with the business cycle view, such as Jacklin and Bhattacharya (1988), Chari and Jagannathan (1988), Gorton (1988) and Allen and Gale (1998). Calomiris and Kahn (1991) consider bank runs as a result of some depositors gathering information about economic fundamentals that would allow assessing the viability of the bank. If the informed depositors conclude that a bank is in trouble, they will withdraw their money and precipitate a run. The sudden withdrawals will force the bank to liquidate all of its assets.

The models of Chen (1999) and Chen and Hassan (2008) may be understood as being related to both the sunspot view, because of the role of first-come, first-served; and to the business cycle view, more particularly to a strand of the business cycle explanation that emphasizes the role of information asymmetry in triggering runs. Chen (1999) shows that failures of a few banks may serve as a noisy signal to depositors who are not able to value their own bank's assets. These uninformed depositors respond to this

negative signal by withdrawing. Since uninformed depositors will withdraw early, the informed depositors are compelled to do the same, even though it would be otherwise better to wait for more precise bank-specific information. Chen and Hassan (2008) extend Chen (1999) to show that expectations about the quality and amount of bank-specific information that will be revealed can affect depositors' incentives to withdraw. Specifically, panic runs can be triggered not only by bank-specific information, but by depositors running when they expect that more noisy information about banks will be revealed, or when they expect that precise information about banks will not be revealed. In their model, panic runs are more likely to occur when the banking industry is weaker. Therefore, we argue that a possible that a possible explanation for the reaction observed in Brazil is that the noisy signal originating from the international financial turmoil increased the uncertainty and fears about the health of the domestic banking system.

The second question is the core of our paper: to understand why there was a flight of deposits from the smaller banks to the big banks. One possible explanation is that depositors thought the largest banks were too big to fail. This is a plausible interpretation in face of the international context and also of the recent history of Brazil. As mentioned before, Lehman's demise caused turmoil in the global financial system and led to a long list of bailouts with Citibank, Bank of America, AIG, ABN Amro, Royal Bank of Scotland, and others. It would be reasonable to think that the Brazilian Government would act in accordance to US and Europe and bailout the big banks should there be any trouble. Moreover, although Brazil does not have an explicit list of too big to fail banks, the country's recent history implied such policy. After the inflation stabilization in 1994, several banks were not able to adjust to the new environment and became distressed. In order to address the banking system fragility, the Brazilian Government launched three major official bank restructuring programs that included government capital injections to the largest private and federally-owned banks of the country. Small private and state-owned banks were allowed to fail. The perception of a too big to fail policy from depositors may have come from the observation of such policy taking place in the US and Europe and from the Brazilian recent history.

To interpret the run to the largest banks as a result of the perception by depositors of an implicit too big to fail policy may be unwarranted, though. An alternative explanation could be that those big banks were in a better shape than the smaller ones. In this case, the run would be in line with the business-cycle view.

There is wide empirical evidence supporting the business-cycle view, such as Calomiris and Gorton (1991), Gorton (1988), Mishkin (1991) and Calomiris and Mason (2003). In our context, one possible interpretation for deposits flying to the big banks in the midst of the crisis would be that depositors were running from banks with worse fundamentals. We take into account this alternative explanation by using several control variables related to bank fundamentals. The first set of variables includes traditional measures of bank risk: equity ratio, the ratio of low quality loans to assets, and the ratio of liquid assets to total assets. A second set of variables aims to control for the effects of the financial crisis in the Brazilian economy. As mentioned before, the worst period of the crisis tightened external financing conditions, so one could argue that bank's dependence of foreign capital to fund its assets became an important risk factor when the financial crisis reached its peak. Also, we mentioned that the crisis led to a decline in international trade that slowed down the Brazilian economy, so it is plausible that depositors could be concerned with particular types of assets that banks held on their balance sheets. For instance, loans to middle market firms may be perceived as particularly risky if these firms have a higher probability of being financially distressed during the economic downturn. We control for the effects of the financial crisis in several robustness checks detailed in section IV.

Finally, to understand why there was a flight of funds from the smaller banks to the big banks during the crisis, it is necessary to control for the other features of size that may be seen as beneficial to depositors. For instance, larger banks are usually more diversified, either by having a large customer base or offering a wide array of financial services and products. There may be other features, such as depositors thinking that larger banks have more cutting-edge technology and risk management techniques. We use the continuous variable log of assets to control for these features.

One potential limitation of this strategy comes from the possibility that depositors value the size discontinuity of the big banks for reasons other than being too big to fail. For instance, depositors might think that big banks are safer because they are widely known, better managed, less subject to information asymmetries and perhaps more closely monitored by the Central Bank. If depositors value the size discontinuity for these other subjective reasons, our interpretation of the too big to fail perception would be confounded. However, these hypotheses do not seem very plausible for at least three reasons. First, the subjective perceptions of safety should be less important for sophisticated institutional investors and our results show that they are precisely the type of depositor who ran more heavily to the big banks. Second, several banks went down despite official supervision from the Fed, FDIC, FSA, etc. Third, as far as monitoring is concerned, Brazilian prudential regulation allows no distinction between the largest banks and other banks.

# A. Identification and Empirical Strategy

To analyze if the anecdotal observation of deposit concentration in the hands of big banks during the crisis is consistent with depositors' perception of a too big to fail policy, we estimate five models with changes in deposits on the left hand size: for uninsured deposits, total deposits, certificates of deposits held by institutional investors, non-financial firms and individual investors. This allows us to examine potential differences in behavior by type of depositors.

To disentangle the other potential benefits of size other than government protection calls for a discontinuity approach, applied by using as one right hand side variable the interaction of an indicator of the worst stage of the financial crisis, (*Crisis*) and an indicator that the bank is part of the set of big banks, (*BigBank*) (the selection of these banks is further detailed in section II-B), together with appropriate controls. The test for the perception of a too big to fail policy during the crisis consists of estimating the sensitivity of the change in banks' deposits to this variable. In other words, the perception of a too big to fail policy during the crisis is tested by estimating the coefficient  $\omega$  of *BigBank* × *Crisis*, our main variable of interest.

The baseline specification is:

$$\begin{split} \Delta Deposits_{i,t} &= \alpha + \phi \Delta Deposits_{i,t-1} \\ &+ \lambda \Delta Premium \ Paid \ on \ Deposits_{i,t} + \beta' Risk_{i,t-1} \\ &+ \delta Regional \ Economic \ Activity_{i,t} + \theta Size_{i,t} + \vartheta Crisis_t \\ &+ \gamma BigBank_i + \omega (BigBank_i \times Crisis_t) + \tau' (Risk_{i,t-1} \times Crisis_t) \\ &+ \eta MechChange_{i,t} + \mu_i + d_t + \varepsilon_{i,t} \end{split}$$

(1)

Where  $\mu_i$  represents the *i*-th bank's time invariant unobserved features that might influence the change in deposits,  $d_t$  stands for time fixed effects (i.e., the common effect of any shock to  $\Delta Deposits$  in time t) and  $\varepsilon$  is the error term.

On the right hand side we have the lagged dependent variable ( $\Delta Deposits$ ), to account for possible momentum or mean reversion effects in the dynamics of the change in deposits; the change in interest rate premium paid on deposits  $(\Delta Premium Paid on Deposits)$ , to account for possible joint determination with change in deposits and avoid an omitted variable bias; a vector of bank fundamentals traditionally found in the literature (Risk), the growth in retail sales, as a proxy for regional economic activity (Regional Economic Activity); the natural logarithm of the assets of the bank (Size), to disentangle the effect of being a big bank from the other features continuously related to size, such as brand equity and convenience; an indicator of the global financial crisis (Crisis); along with the necessary interactions. The model specification also deals with a regulatory change of the amount of insured in August, 2006, with a deterministic variable, MechChange. All variables used in this study are formally described in the Appendix.

The variables used in the *Risk* vector are: the equity ratio, to evaluate leverage (*Equity ratio*); the ratio of low quality loans to assets, to evaluate the risk of the loan portfolio (*Low Quality Loans*); and the ratio of liquid assets to total assets, to assess liquidity risk (*Liquidity*).

Table 4 presents the estimation results of this baseline specification for uninsured and total deposits and Table 12 for institutional investors, non-financial firms and individual investors.

The models are estimated using Pooled Ordinary Least Squares (POLS) and the system Generalized Method of Moments (GMM-Sys) procedure described in Blundell and Bond (1998).

The system GMM allows us to explicitly model the bank unobserved fixed effect represented by  $\mu_i$  and consistently include the lagged dependent variable among the regressors, unlike other panel data estimators, such as the traditional Fixed Effects and Random Effects procedures. In addition, GMM-Sys enables us to deal with the plausible endogenous relationship between bank fundamentals, change in interest rate premium, and change in deposits, by using suitable lagged values of the regressors as instrumental variables. A similar procedure permits us to tackle the issue of dynamic endogeneity (e.g., see Wintoki, Linck and Netter, 2010) caused by the potential influence of shocks to the change in deposits over bank fundamentals in future periods (e.g., governance or management changes that affect deposits contemporaneously and the bank risk profile in subsequent periods). The plausibility of our identifying assumptions (i.e., the appropriateness of the set of lagged variables that we choose as instruments) is formally tested by the Hansen/Sargan test of overidentifying restrictions and the Arellano-Bond test for error autocorrelation. In all regressions reported in Table 4 we cannot reject the null hypotheses, suggesting that our identifying assumptions are acceptable. Finally, with both POLS and GMM-Sys we control for time fixed effects by using time dummy variables.

As mentioned before, we also estimate other seven specifications to take control for alternative explanations of the run that are detailed in Section IV - Robustness checks. Tables 5 to 11 present the estimation results of these alternative specifications.

# B. Big Banks

Systemic risk can be defined as a risk of disruption to financial services caused by a deterioration of all or parts of the financial system that potentially brings a negative impact on the real economy. Institutions that pose a systemic risk are usually referred to as "too big to fail". The term was first used in a congressional hearing in 1984, right after the bailout of Continental Illinois, when the regulator of US national banks testified that 11 of the largest banks would receive a similar treatment if necessary (Mishkin, 2006). At that time, size of assets was the criterion to characterize systemically important financial institutions.

In our model, the identification of the set of banks that could be perceived as too big to fail is based on their importance in the Brazilian market, combining outstanding size, substitutability and interconnectedness. Only domestic banks, including those with foreign control, were considered as candidates. Our definition of big banks is consistent with the concept of systemically important financial institutions. We base our selection procedure on several cluster analyses and multidimensional scaling graphs. For these formal data analyses we use five variables: (i) total assets plus brokerage, (ii) total assets, (iii) total deposits, (iv) number of branches, and (v) number of clients.

Table 1 shows the classification suggested by two clustering algorithms, known as *K*-means and *K*-medians (for details, see Kaufman and Rousseeuw, 2005). In both cases we specify the number of nonoverlapping groups (k) to be formed through an iterative process. In other words, we determine the number of groups to be formed and the algorithm chooses how many banks and which banks to assign to each group. Specifically, the *K*-means procedure assigns each bank to the group whose (multivariate) mean is closest, whereas the *K*-medians does the same, but using medians instead of means to represent the group centers. The algorithms begin with k randomly chosen seed values, which act as the k group means or medians in the first step. Then, based on the initial categorization, new group means/medians are computed. This procedure is repeated until no observations change groups. Table 1 shows that, when k = 2, the first cluster is composed by eight banks that are distinctively larger (in terms of the five variables we employ) than the remaining banks. The result is the same with both algorithms.

# [Insert Table 1 here]

To visualize these clustering patterns we perform a classical multidimensional scaling analysis, which is a technique that allows us to represent high-dimensional space dissimilarities between observations in a lower-dimensional space. Specifically, we use the Euclidean distance between banks in the two-dimensional space to approximate the actual distances in the five-dimensional space (since we use five variables). The multidimensional scaling configuration graph is shown in Figure 4. Figure 4 reveals that the eight banks selected by the cluster analysis are somewhat distinct from other banks, taking into consideration the five dimensions of systemic importance defined above.

# [Insert Figure 4 here]

# III. Data and Sample Selection

The Brazilian Financial System has several distinguishing features that have helped shape the industrial organization of that country. First, the banking system is regulated and supervised exclusively at the federal level. The Central Bank of Brazil and the National Monetary Council<sup>4</sup> (CMN, for its acronym in Portuguese) are in charge of regulation and the Central Bank is also in charge of authorizations and supervision.

<sup>&</sup>lt;sup>4</sup> The National Monetary Council is formed by the President of the Central Bank and two State Secretaries.

Second, the banking industry is formed mostly of universal banks. It has been like that since the end of the 1980s, when there was a regulatory change allowing banks to offer different services such as commercial banking, investment banking, consumer financing and so on.

Third, as mentioned before, banks are allowed to open branches anywhere, by discretion of the Central Bank of Brazil. That is, there are no rules forbidding interstate or international branching, but the Central Bank has to authorize each and every opening.

Fourth, in certain aspects, regulatory restrictions are more stringent than in developed economies. For instance, Brazilian banks are required to have a capital adequacy ratio of at least 11%, larger than the 8% Basle requirement. Furthermore, tier II capital may not exceed tier I capital. Also, the Central Bank operates a real-time gross settlement (RTGS) payment system since April 2002, which avoids the possibility of overdrafts in reserves at any time.

The Brazilian Deposit Insurance Fund, Fundo Garantidor de Crédito (FGC, for its acronym in Portuguese) is mandatory, privately-funded and was founded in 1995, in the aftermath of the restructuring programs mentioned before. The FGC covers the amount held by each person against one financial conglomerate up to BRL 60,000 – around USD 30,000. Initially it covered up to BRL 20,000 but the amount of coverage was extended in June 2006, during a calm period for the Brazilian banking system.

Almost all types of deposits in Brazil are in local currency and eligible for deposit insurance<sup>5</sup>. The distribution of deposits is shown on Panel A of Table 2. Checking account deposits, savings deposits and time deposits accounted for around 93% of the funds deposited in Brazilian banks during the sample period. The remaining 7% include interbank deposits and other deposits, such as those related to litigations. Checking account deposits pay no interest and, on average, accounted for 16% of total deposits over the sample period. Savings deposits pay an interest rate determined by law: a floating interest rate of 6 percentage points over a fraction of a specific inflation index. On average, savings deposits accounted for 28% of total deposits. Checking and savings

<sup>&</sup>lt;sup>5</sup> Deposits in foreign currency account for less than 2% of total deposits in Brazil and are allowed only to very specific types of investors (non-resident persons and companies). Some other types of deposits are not eligible for deposit insurance, such as: i) deposits, loans or any other type of funding raised abroad; ii) deposits related to litigations; iii) time deposits authorized to compose Tier-2 of the regulatory capital. These ineligible-for-insurance deposits account for less than 10% of the overall amount of deposits in the Brazilian Financial system.

deposits may be withdrawn on demand by the customer without notice or penalty. Time deposits (mostly certificates of deposits) pay interest, and the rate may be fixed or floating. Around 70% of the time deposits in Brazil allow early withdrawal (eventually with a penalty rate). Throughout the sample period, time deposits accounted for 49% of total deposits on average.

Our primary database consists of semiannual observations of all deposit-taking banks in Brazil in the database of the Central Bank of Brazil between December/2001 and December/2009 (17 periods). We exclude from our sample banks that do not appear among the top 50 in either deposit taking or total assets in any of the 17 periods. We also exclude banks that were under Central Bank intervention. We require that the ratio of deposits over assets be higher than 1% and that all observations have nonmissing data for book assets, while all multivariate analysis implicitly requires nonmissing data for the relevant variables. To mitigate the impact of data errors and outliers on our analysis, we Winsorize all variables at the 5<sup>th</sup> and 95<sup>th</sup> percentiles.

Panel B of Table 2 lays out the representativeness of the sample. We have a total of 74 banks in the beginning of the sample period, and end up with 53 banks.(the number decreases over time due to mergers and acquisitions and only one bank failure occurred in 2004). This sample of banks hold from 96.1 to 99.1% of the deposits eligible for deposit insurance in the Brazilian Financial System along the studied period. Panel B also shows a large increase in the amount of deposits holdings by Brazilian banks: in less than 8 years, the amount of deposits has increased fourfold in Brazil. During the same period, the cumulative inflation rate was almost 70%. This rise can be attributed to a series of factors, such as nominal GDP growth of 141% in the sample period, the sharp increase in credit operations, the inclusion of the lower classes of the population into the banking system and the maintenance of high interest rates by the Central Bank in the period.

# [Insert Table 2 here]

The data we use in this study has three sources. The first set of data is available to the public, provided by the Central Bank of Brazil. It comprises detailed balance sheet, income and earnings reports, as well as data on the number and location of branches, and selected regulatory indicators, such as the capital adequacy ratio for all Brazilian banking firms.

The second source for our data is private, and comes from the Brazilian Deposit Insurance Fund, (Fundo Garantidor de Crédito – FGC). This is a unique bank level data on the number of depositors and volume of deposits in several different deposit-size brackets for all Brazilian banking firms. This novel data allow us to compute the volume of insured and uninsured deposits of each bank in each period<sup>6</sup>.

Third, we use private data provided by the Central Bank of Brazil that comprises daily balances of certificates of deposits in the hands of institutional investors, non-financial firms and individual investors; and semiannual information on the different types of bank loans outstanding.

Fourth, we also use data from the retail sales index provided the Brazilian Institute for Geography and Statistics (IBGE, for its acronym in Portuguese), which provides the growth in retail sales for each state of the federation<sup>7</sup> as well as the resulting national growth in retail sales. This is the most used indicator of regional economic activity in Brazil.

We treat merged banks (or acquisitions in which two different banks start consolidating their balance sheets) as new banking entities. For example, if Bank A acquires Bank B (or even if Bank A and Bank B merge into bank AB), we treat the merged bank as a new bank, Bank C. There were two mergers among the largest banks in Brazil during the exacerbation of the global financial crisis, in the end of 2008. In this case, the change in deposits was calculated based on the sum of deposits of the two merging banks.

# A. Summary Statistics

Summary statistics are shown in Table 3 for dates Dec/2001, Jun/2008 and Dec/2009. We split the statistics into big banks and other banks.

# [Insert Table 3 here]

Overall, banks have notably increased asset size and equity during the sample period. The other banks rely more on uninsured deposits than the big banks, which is explained by the fact that big banks have more branches and hold deposits of small retail clients, while the other banks rely mostly on middle market, corporate and institutional

<sup>&</sup>lt;sup>6</sup> The periods range from January 1<sup>st</sup> to June 30<sup>th</sup> and July 1<sup>st</sup> to December 31<sup>st</sup>.

<sup>&</sup>lt;sup>7</sup> Brazil has 27 states.

depositors. In addition, big banks have a slightly higher ratio of low quality loans to assets and lower equity ratio on average. In the beginning of the sample period, big banks had slightly higher liquidity than the other banks. In the period immediately before the crisis, we note that both sets of banks had experienced a decrease in liquidity, especially big banks, which ended up less liquid than the other banks. The same pattern remains in the last sample period. The most striking difference between big banks and the other banks is deposit concentration (the portion of a bank's asset being funded by each depositor on average). The ratio of deposit concentration of other banks to big banks is over 9,800 in December/2009.

# **IV.** Results

Our main parameter of interest is the coefficient ( $\omega$ ) of the interaction of the global financial crisis dummy with our big bank dummy (*BigBank* × *Crisis*). It captures the expected difference in the percent change in bank deposits between big banks and other banks during the most critical stage of the financial crisis, controlling for fundamentals, change in interest rate paid on deposits, size of assets and macro effects. In other words, a positive and significant  $\omega$  in the uninsured deposits regression indicates that depositors behave consistently with the perception of a too big to fail policy. In contrast, we expect a lower  $\omega$  in the total deposits regression.

The results of the estimations of the models of uninsured and total deposits, using Pooled Ordinary Least Squares (POLS) and the System Generalized Method of Moments (GMM-Sys) are shown in Table 4.

# [Insert Table 4 here]

We find a positive and statistically significant  $\omega$  for the deposits regressions under both estimation procedures (in all cases, at the 1% level). These estimates are also economically large, corresponding to a predicted increase of approximately 45 percentage points in uninsured deposits for the big banks group in comparison with other banks during the critical stage of the crisis. In addition, as expected, we find a positive but much lower  $\omega$  for total deposits regression under all estimation procedures (predicting approximately 36 percentage point additional increase in deposits for big banks during the crisis).

The coefficient of the *Big bank* dummy is not statistically significant, at conventional levels, in any regression. This means that, during normal times, the percentage change

of both uninsured and total deposits is unrelated to whether the bank is in our big bank group or not. Thus, our evidence suggests that investors might perceive a too big to fail policy and that such perception is relevant only in times of crisis.

The coefficient of *Size* is nonsignificant in the uninsured and total deposits models, no matter the estimation procedure. All of the other controls shown in Table 4 (equity ratio, low quality loans and liquidity) and their interactions with the crisis dummy have non-significant coefficient estimates. These widely used proxies for bank fundamentals are not relevant for explaining the behavior of depositors neither in normal times nor during the financial crisis. The results shown in Table 4 also show that interest rates paid on deposits have little or no power to explain the change in deposits, which is consistent with the idea that money markets are risk intolerant. The fact that traditional control variables have little explanatory power is intriguing, because it suggests that depositors are not sensitive to bank fundamentals both in normal times and during the crisis. Since previous empirical research has found evidence supporting runs based on fundamentals, we believe it is very likely that one or more important risk factors related to the characteristics of the crisis may be missing in this specification. We shed some light on the subject in section IV.C below.

Overall, the estimates shown in Table 4 suggest that the positive spike in deposits of big banks during the financial crisis cannot be explained by the heterogeneity in bank fundamentals, by a simple size effect or by a general propensity of such banks to attract more deposits than their competitors.

# V. Robustness Checks

# A - Isolating the impacts of the Global Financial Crisis on the Brazilian Financial System

# A.1. - Types of loans

During the most critical stage of the global financial crisis, the set of big banks may have been considered safer not necessarily because they would be bailed out, but because depositors believe those big banks would be more resilient to the crisis effects. We take a deeper look at bank loans, since it is plausible that depositors could be concerned with particular types of assets that banks were exposed to on their balance sheets. We are especially interested in banks that engage in trade finance loans and middle market operations (loans made to small and medium-sized firms). Trade finance loans are very collateralized by import/export contracts, have typically very low delinquency rate and loss given default, and are thus expected to be safe during the crisis. On the other hand, middle market loans have typically low collateral and are held to maturity by the lender bank (instead of being securitized and traded in the secondary market) and thus we could expect depositors to percept these loans to be riskier during the financial crisis, since small and medium firms have a higher probability of being financially distressed during the economic downturn.

# [Insert Table 5 here]

The results shown in Table 5 show that, although the signs of the coefficients estimates for the interactions of these types of asset exposure with the crisis dummy (*trade finance x crisis* and *middle market x crisis*) are consistent with the above rationale, they are not statistically significant at the usual levels. The other coefficient estimates of interest (especially *big bank x crisis*) are practically unchanged by the inclusion of these variables.

A.2 – Liquidity freeze

# [Insert Table 6 here]

As mentioned before, one of the effects of the financial turmoil was tight external financing conditions. After Lehman's bankruptcy the interbank market virtually froze, so banks that depended more heavily on external financing would probably suffer the most and could become distressed. We use the ratio of foreign funds to total assets as a proxy for bank's dependence on external financing on the right hand side of our model. However, we find no evidence that depositors favored banks with lower dependence on foreign funding, as we show on Table 6.

A.3 – Deposit concentration and reliance on institutional depositors

Another indirect measure of exposure on the liabilities side is depositor concentration, measured as the fraction of assets being funded by each depositor on average. Banks that have a narrower depositor base, where few depositors hold a large share of the total deposits, may be in impending distress. For instance, if some of these depositors had to withdraw their funds at the same time due to liquidity reasons related to the global financial crisis, the bank might lose a significant share of its funding. This would be exacerbated under the Goldstein and Pauzner (2005) model, that states that when noisy information is revealed, depositors would tend to run because the bank could be in

trouble if only a few depositors decide to run first. Therefore, in order to control for this feature, we use the natural logarithm of the ratio of the average deposit size to total assets as a right hand side variable that accounts for deposit concentration. We use logs to mitigate the extreme right-tail asymmetry of this variable. Results shown in Table 7 indicate that there is no evidence that depositor concentration affects the growth in deposits.

# [Insert Table 7 here]

Under the rationale above, the first-come, first-serve issue is even more exacerbated if depositors assume one (or both) of the two hypotheses: i) that other depositors have superior information; ii) that other depositors are extremely risk-averse and thus will run when noisy information is expected. Institutional investors (such as pension and mutual funds) are the typical case of well informed and, in some cases, risk-averse depositors. As mentioned before, Chen, Goldstein and Jiang (2010) find evidence that the behavior of institutional investors depends on whether they are surrounded by other institutional investors or by retail investors in mutual funds. To account for this factor, we use a measure of reliance on institutional investors for funding, which is the ratio between the amount of certificates of deposit held by institutional investors and total assets. Table 8 shows that, during normal times, the reliance on institutional investors is not important for driving deposits' growth, while there is a very significant (economic and statistical) negative effect of relying on deposits of institutional investors during the financial crisis for both uninsured and total deposits, which is consistent with Chen, Goldstein and Jiang (2010) empirical findings. During the financial crisis, a 1 percentage point increase in the reliance on institutional investors would decrease deposits by over 2.8 percentage points, for both uninsured and total deposits. In addition, when reliance on institutional investors is considered in our regressions, the coefficient of the interaction of liquidity and crisis turns out to show a positive and significant (at the 10% level) effect on deposits. The inclusion of these additional controls slightly changes the magnitude and standard errors of some coefficients but it does not significantly alter our inferences. In particular, the estimates for the *big bank x crisis* interaction lowers to the 38-48 percentage points range for uninsured deposits and 29-38 percentage points range for total deposits. The results in Table 8, however, could simply mean that institutional investors were the ones who ran from deposits, so that the higher the concentration of deposits held by institutional investors in a certain bank, the more it lost deposits (or the

less it gained deposits). We show that this was not the case, when we return to this issue in section C below.

# [Insert Table 8 here]

# B – Government-owned banks

We then investigate if depositors favor government-owned banks against privately-held banks. 10 banks in our sample are controlled by the government (4 by the Federal Government and 6 by states of the federation), 2 of which are included in our list of big banks. It is plausible to assume that deposits perceive government-owned banks as enjoying some kind of implicit guarantee to depositors, so that these banks could be considered *too-protected-to-fail*. To account for this possibility, we include a dummy variable for non-big banks government-owned banks (i.e. banks that are controlled by the government and do not belong to our list of big banks) and also interact it with the financial crisis dummy. The results shown in Table 9 indicate that these banks do not enjoy higher deposit growth during normal times. The coefficient of the interaction variable (*government-owned bank x crisis*) shows some evidence, although weak, that the increase in both uninsured and total deposits during the financial crisis was higher for government-owned banks.

# [Insert Table 9 here]

To further investigate the issue of an implicit protection to government-owned banks, we investigate whether the *big bank x crisis* effect was larger for the 2 government-owned banks of our big bank group than for privately-owned banks in the group. We do that by first excluding big private banks from the sample (results in Table 10) and then excluding big government-owned banks (results in Table 11). The results of Table 10 and 11 show that there is virtually no difference in our estimates of the *big bank x crisis* coefficient for uninsured deposits between private and government-owned banks (either of them enjoy circa 45 percentage points more growth in uninsured deposits compared to the other banks). When we turn our attention to total deposits, the results also show little difference: we observe a coefficient of 34 percentage points for government-owned banks and 38 percentage points for private banks.

# [Insert Table 10 here]

[Insert Table 11 here]

# *C* - *The behavior of different types of depositors*

We then investigate the behavior of different types of depositors. Certificates of deposits may be held by each of the three classes of investors: 1) institutional; 2) non-financial firms and; 3) individuals. It would be reasonable to conjecture that the first are the ones with superior information and higher degree of sophistication among the 3 classes, while individuals would be less informed and less sophisticated. We then run our baseline specification for each of the different classes of investors.

The results of Table 12 show that, during normal times, institutional investors are sensitive to banks' equity ratio. Our estimates imply that a 1 percentage point increase in the equity ratio predicts an increase in the growth rate of certificates of deposit held by institutional investors of around 1.1 percentage points. Institutional investors are sensitive to banks' exposure to trade finance during normal times and during the crisis. In normal times, a greater exposure is mildly penalized by institutional investors: a 1 percentage point increase in exposure predicts a fall in the growth rate of deposits of 0.11 percentage point. However, during the financial crises, the same increase of 1 percentage point in exposure predicts a rise in the growth rate of deposits of 0.74 percentage points. One possible explanation is that, in normal times, a higher diversification of the loan portfolio is preferred, but during the crisis, banks that have higher concentration on the safer types of loans are better off.

Regarding our main variable of interest, we find very large coefficients for institutional investors although with different degrees of statistical significance, depending on the model specification (2 at the 5% level, 1 at the 1% level and 1 at the 10% level). For non-financial firms coefficients are always significant at the 1% level, but smaller in magnitude. In columns (4) and (8), when we control for the presence of other institutional investors, the magnitude of the coefficients of the *big bank x crisis* substantially lowers for both types of investors. These findings suggest that some part of the run could be explained by incentives to withdraw depending on whether depositors fear others will withdraw first. The estimates shown in columns (9) to (12) suggest that individual investors did not favor big banks during the crisis.

# D. The post-crisis period

The previous tests have shown that depositors favored big banks, banks that relied less on the funding of institutional investors during the crisis and that these results were driven by the behavior of both institutional investors and non-financial corporations. We now investigate how deposit growth evolved for different banks in the post-crisis period, which we define as the first and second semesters of 2009 (Jun-09 and Dec-09). Both the OLS and the GMM regressions reported on table 13 show that the change in uninsured deposits during the post-crisis period (*big bank x post crisis*) for other banks was 15 percentage points larger compared to big banks (13% for total deposits), with 10% statistical significance. It is important to note that the coefficient obtained for the post crisis period (by itself, not interacted with big bank) is not significantly different from zero at the usual levels. In addition, the coefficients of *reliance on institutional investors x post crisis* show that banks that had a larger share of their assets funded by CDs of institutional investors experienced larger deposit growth (for both uninsured and total deposits, under OLS and GMM specifications) in the post crisis period.

# [Insert Table 13 here]

These results reveal important information on the behavior of depositors. Indeed, there is evidence that the run identified in other banks during the critical stage of the financial crisis is partially reverted for uninsured deposits and almost entirely reverted for total deposits in the post crisis period (note that there are 2 periods considered *post crisis*). This evidence is in some sense consistent with the hypotheses that relate bank runs to noisy information that is revealed (or expected) during crises and the too big to fail hypothesis.

# [Insert Table 14 here]

We also look at how different types of depositors (institutional investors, non financial companies and individuals) behaved after the crisis. The results in Table 14 show that the coefficients for *big bank x post crisis* are negative and statistically and economically significant for institutional investors under all specifications, whilst the coefficients of the other variables of interest (including *big bank x crisis*) remain practically unchanged. In fact, the positive change in deposits of institutional investors observed for big banks during the crisis is more than reverted after the crisis<sup>8</sup>. For non financial firms and individuals, we also observe negative coefficients for *big bank x post crisis*, but in only one case it is statistically significant at the 10% level. These results clearly indicate

<sup>&</sup>lt;sup>8</sup> Again note that our definition of post crisis includes 2 semesters, and the coefficients of *big bank x.post crisis* is more than half the coefficient of *big bank x crisis*.

that institutional investors performed a "flight-to-big-banks" movement during the crisis (and **only** during the crisis).

This reversal in the change in deposits to other banks observed in the post crisis period can also be attributed, at least in part, to the creation of a special CD with guarantee of up to BRL20 million (around USD 9 million) in March 2009 as mentioned before. In fact, there is anecdotal evidence reporting that institutional investors account for a great portion of this special guaranteed CD.

# E. Additional controls for bank fundamentals

We implement several other robustness tests to check the stability of our main results. Our baseline specification uses bank fundamentals in level, in accordance to the market discipline literature. However, it is possible that depositors are indeed interested in trends of bank fundamentals. Thus, we substitute  $\Delta R_{i,t-1}$  for  $R_{i,t-1}$  in equation (1), meaning that we now control for the *change* in bank fundamentals (i.e., capital adequacy, asset quality, management quality, earnings, and liquidity) from t - 2 to t - 1. This alternative specification addresses the possibility that clients are mainly sensitive to improvements or deteriorations of bank fundamentals instead of their level when deciding to withdraw or expand their deposits. In these regressions (results not reported), the coefficients estimated for  $\Delta R$  turn out to be nonsignificant in all cases. However, our coefficient of interest ( $\omega$  in equation (1)) again remains practically unchanged.

# F. Alternative estimators and identifying assumptions

We also check our results employing alternative estimators and/or identifying assumptions. First, we use the GMM fixed effects panel data estimator proposed by Arellano and Bond (1991) with identifying assumptions regarding the endogeneity of some regressors similar to those used in the GMM regressions reported in Table 4, namely, allowing the bank fundamentals contained in vector *R*, plus  $\Delta Deposits$ ,  $\Delta IntRatePremium$ , and *Size* to be only sequentially exogenous (i.e., potentially correlated with the error term  $\varepsilon$  in some time periods). Specifically, by using suitable lagged values as instruments, we let bank fundamentals and size to be correlated with past shocks, thus allowing for feedback effects running from the change in deposits to those variables. Similarly, we let  $\Delta Deposits$  and  $\Delta IntRatePremium$  to be correlated with past as well as contemporaneous values of  $\varepsilon$ , thus accounting for the likely

simultaneous determination of the volume and price of deposits. We also rerun all regressions using the two-step GMM estimator instead of the one-step procedure reported in Table 4. Finally, we employ alternative identifying assumptions, such as allowing the bank fundamentals to be correlated with  $\varepsilon$  contemporaneously, as well. For the benefit of space, we do not report the results of these exercises. In all cases, our main inferences are not materially affected, though.

# VI. Concluding remarks

When a too big to fail policy is in place, either formally or implicitly, there are fewer incentives for debt holders and depositors to discipline financial institutions, leading to an increase in moral hazard.

This study shows that the perception of an implicit too big to fail policy affects depositor behavior. Specifically, we analyze the reaction of Brazilian depositors to the international financial turmoil triggered by Lehman Brother's demise in September 2008. Taken together, our results indicate that depositors moved their funds to the largest banks of the country because depositors thought those banks would not be allowed to fail. The effect for uninsured deposits was larger than for total deposits. Likewise, the effect for certificates of deposits held by institutional investors was larger than for those held by non-financial firms. We find that depositors' response to bank fundamentals was relatively weak, which seem to support the view that a run may occur when depositors learn information from other banking systems, which serves as a noisy signal on domestic bank-specific information. Depositors feared for their funds and decided to run to where they believed was safe: the largest banks of the country, implicitly protected by the Federal Government and the Central Bank. This result is in line with some features of the bank runs model of Chen (1999) and Chen and Hassan (2008).

We also find that the presence of institutional investors seems to have played a role in the run in Brazil. Banks that depended on institutional investors for funding suffered more deposit outflows, not only from institutional investors themselves, but also from non-financial firms. This result indicates that some part of the run can be explained by incentives to withdraw depending on whether depositors fear others will withdraw first, which is in line with the evidence from mutual funds outflow found in Chen, Goldstein and Jiang (2010) and with some features of the model for bank runs of Goldstein and Pauzner (2005).

Taken together, our results indicate that, in the face of uncertainty, depositors considered the largest banks too big to fail and behaved consistently. This calls our attention to possible cross-border spillover effects caused by public policies in developed markets and to the need for improved market discipline. It also makes a strong case to the need of an international approach to bank resolution.

# References

- Allen, Franklin, Ana Babus and Elena Carletti, 2009, Financial crises: Theory and evidence, *Annual Review of Financial Economics*, 1, 97–116.
- Allen, Franklin and Elena Carletti, 2010, An Overview of the Crisis: Causes, Consequences, and Solutions, *International Review of Finance*, vol. 10, issue s1, pages 1-26.
- Allen, Franklin and Douglas Gale, 1998, Optimal Financial Crises, *Journal of Finance* 53, 1245-1284.
- Arellano, Manuel and Stephen R. Bond, 1991, Some tests of specification for panel data: Monte Carlo evidence and an application to employment equations, *Review of Economic Studies*, 58, 277-297.
- Beltratti, Andrea and Stulz, René M. Why did some banks perform better during the credit crisis? A cross-country study of the impact of governance and regulation. *NBER Working Paper* w15810. Available at <u>http://www.nber.org/papers/w15180</u>, July 2009.
- Bevilaqua, Afonso S. and Marcio G. P. Garcia, 1999, Banks, Domestic Debt Intermediation and Confidence Crises: the Recent Brazilian Experience, Available at SSRN: http://ssrn.com/abstract=228783 or doi:10.2139/ssrn.228783.
- Blundell, Richard and Stephen R. Bond, 1998, Initial conditions and moment restrictions in dynamic panel data models, *Journal of Econometrics*, 87, 115-143.
- Boyd, John H. and David E. Runkle, 1993, Size and Performance of Banking Firms: Testing the Predictions of Theory, *Journal of Monetary Economics*, 31(1), 47–67.
- Brewer, III, Elijah and Julapa Jagtiani, 2009, How much did banks pay to become toobig-to-fail and to become systemically important?, Working Papers 09-34, *Federal Reserve Bank of Philadelphia*.
- Brunnermeier, Markus K., 2009, Deciphering the Liquidity and Credit Crunch 2007–2008, *Journal of Economic Perspectives*, Volume 23, Number 1, Pages 77–100.
- Calomiris, Charles W. and Gary Gorton, 1991, The origins of banking panics, models, facts, and bank regulation, in R. G. Hubbard, ed., *Financial Markets and Financial Crises*, Chicago, IL: University of Chicago Press.
- Calomiris, Charles W. and Charles M. Kahn, 1991, The role of demandable debt in structuring optimal banking arrangements, *American Economic Review*, 81, 497-513.
- Chen, Qi, Itay Goldstein and Wei Jiang, 2010, Payoff complementarities and financial fragility: Evidence from mutual fund outflows, *Journal of Financial Economic*, 97, 239-262.
- Chen, Yehning, 1999, Banking panics: the role of first-come, first-served rule and information externalities. *Journal of Political Economy*, 107, 5: 946-968.
- Chen, Yehning and Iftekhar Hasan, 2008, Why do bank runs look like panic? A new explanation. *Journal of Money, Credit and Banking*, 40: 535–546.
- Cumbo, Josephine, 2008, October 3, Safe savings: Your questions answered, *Financial Times*. Retrieved from: http://www.ft.com.

- DeYoung, Robert, Douglas D. Evanoff and Philip Molyneux, 2009, Mergers and acquisitions of financial institutions: A review of the post-2000 literature, *Journal of Financial Services Research*, 36, 87-110.
- Diamond, Douglas W. and Philip H. Dybvig, 1983, Bank runs, deposit insurance, and liquidity. *Journal of Political Economy*, 91:401–19.
- Goldfajn, Ilan, Katherine Hennings and Helio Mori, 2003, Brazil's Financial System: Resilience to Shocks, No Currency Substitution, but Struggling to Promote Growth, Working Paper No. 170, *Center For Research On Economic Development And Policy Reform.*
- Goodhart, Charles A.E. and Haizhou Huang, 2005, The lender of last resort, *Journal of Banking and Finance*, 29, 1059-1082.
- Gorton, Gary, 1988, "Banking Panics and Business Cycles," Oxford Economic Papers 40, 751-781.
- Gorton, Gary, 2009, Slapped in the Face by the Invisible Hand: Banking and the Panic of 2007 (May 9, 2009). Available at SSRN: <u>http://ssrn.com/abstract=1401882</u>
- Gorton, Gary and Andrew Winton, 2003, Financial Intermediation, in G. Constantinides, M. Harris, and R. Stulz, eds., *Handbook of the Economics of Finance*, Volume 1A, Chapter 8, North Holland, Amsterdam, 431-552.
- Gropp, Reint, Hendrik Hakenes and Isabel Schnabel, 2010, Competition, Risk-Shifting, and Public Bail-out Policies. Preprints of the Max Planck Institute for Research on Collective Goods. Bonn 2010/05. Available at: <u>http://www.coll.mpg.de/pdf\_dat/2010\_05online.pdf</u>.
- Hall, Ben, 2008, October 2, Fillon vows to shield French banks, *Financial Times*. Retrieved from: http://www.ft.com.
- Herring, R., Carmassi, J., 2009, The corporate structure of international financial conglomerates: complexity and implications for safety and soundness, in Berger, A., Molyneux, P., Wilson, J.O.S. (Eds), Oxford Handbook of Banking, Oxford University Press, Oxford,
- Imai, M., 2006, Market discipline and deposit reform insurance in Japan, *Journal of Banking and Finance*, 30, 3433-52.
- International Monetary Fund (IMF). 2009a, Regional economic outlook: Western Hemisphere. Stronger Fundamentals Pay Off. Washington, D.C. May, 2009.
- International Monetary Fund (IMF). 2009b, Regional economic outlook: Western Hemisphere. Crisis Averted What's next? Washington, D.C. October, 2009.
- Ioannidou, Vasso P. and María Fabiana Penas, 2010, Deposit insurance and bank risktaking: Evidence from internal loan ratings, *Journal of Financial Intermediation*, 19, 95-115.
- Jacklin, Charles and Sudipto Bhattacharya, 1988, Distinguishing panics and Information-Based Bank Runs: Welfare and Policy Implications, *Journal of Political Economy* 96, 568-592.
- Kaufman, Leonard and Peter J. Rousseeuw, 2005, *Finding groups in data: an introduction to cluster analysis*, Wiley, New York.

- Keister, Todd, 2010, Bailouts and Financial Fragility, *Federal Reserve Bank of New York*, Staff Reports: 473.
- Krishnan, C. N. V., P. H. Ritchken and J. B. Thomson, 2005, Monitoring and controlling bank risk: Does risky debt help?, *Journal of Finance*, 60 (1), 343-378.
- Maechler, Andrea M. and Kathleen M. McDill, 2006, Dynamic depositor discipline in US banks, *Journal of Banking and Finance*, 30, 1871-1898.
- Mesquita, Mário and Mario Torós, 2010, Considerations on the role of the Central Bank during the 2008 Crisis (In Portuguese: Considerações sobre a atuação do Banco Central na Crise de 2008), Working Paper Series n. 202, p. 1-39.
- Mishkin, Frederic S., 1991, Asymmetric information and financial crises: a historical perspective, in Glenn Hubbard, editor, *Financial Markets and Financial Crises* (University of Chicago Press; Chicago), 69-108.
- Mishkin, Frederic S., 2006, How big a problem is too big to fail? A review of Gary Stern and Ron Feldman's Too big to fail: The hazards of bank bailouts, *Journal of Economic Literature* 44, 988-1004.
- O'Hara, Maureen, and Wayne Shaw, 1990, Deposit insurance and wealth effects: The value of being too big to fail, *Journal of Finance* 45, 1587–1600.
- Park, Sangkyun, and Stavros Peristiani, 1998, Market discipline by thrift depositors, *Journal of Money, Credit and Banking*, 30 (3), 347-364.
- Peria, Maria S. M. and Sergio L. Schmukler, 2001, Do depositors punish banks for band behavior? Market discipline, deposit insurance and banking crises, *Journal of Finance*, 54 (3), 1029-1051.
- Romero, Cristiano and Alex Ribeiro, 2009, Brasil enfrentou corrida bancária na crise. *Jornal Valor Econômico*, Caderno A, ed. 13<sup>th</sup> of November, 2009.
- Ross, Alice, 2008, October 7, Cash inflows into HSBC up, *Financial Times*. Retrieved from: <u>http://www.ft.com</u>.
- Schnabel, Isabel, 2009, The Role of Liquidity and Implicit Guarantees in the German Twin Crisis of 1931, *Journal of International Money and Finance*, 28, 1–25.
- Sloan, Steven, 2009, Moral hazard seen as the price to save system, *American Banker*, 174 (217), 8.
- Taylor, John D., 2009, Economic policy and the financial crisis: an empirical analysis of what went wrong, *Critical Review*, 21 (2-3), 2009, pp. 341-364.
- Vincent, Matthew, 2008, October 3, Government-backed savings: Safety deposits, *Financial Times*. Retrieved from: <u>http://www.ft.com</u>.
- Warwick-Ching, Lucy, 2009, March 3, Record deposits into National Savings, *Financial Times*. Retrieved from: <u>http://www.ft.com</u>.
- Wintoki, M. Babajide, James S. Linck and Jeffry M. Netter, 2010, Endogeneity and the dynamics of internal corporate governance, CELS 2009 4th Annual Conference on Empirical Legal Studies Paper, Available at SSRN: http://ssrn.com/abstract=970986.
- Willman, John, 2008, October 9, Big economies ready to take lead from London, *Financial Times*. Retrieved from: http://www.ft.com.

# Figure 1. Market Share of Certificates of Deposits (CDs)



The purple solid line shows the daily evolution of the market share of CDs of the big banks. The green dotted line shows the daily evolution of the market share of CDs of the other banks.



Figure 2. Total deposits market share of the set of Big Banks

Source: Central Bank of Brazil

# Figure 3. Total Deposits and Real Effective Exchange Rate Index



The blue line shows the evolution of total deposits (in billions of BRL – left axis) and the real effective exchange rate index.

Source: Central Bank of Brazil

# Figure 4. Multidimensional Scaling Configuration Graph

The graph below represents the Euclidian distances between banks in two-dimensional space as an approximation of the original distances computed for the following five variables (in standardized form): (i) total assets plus brokerage, (ii) total assets, (iii) total deposits, (iv) number of branches, and (v) number of clients.


#### Table 1 – Big banks – Results from the cluster analysis

The groups shown in this table were suggested by the cluster analysis algorithms *K*-means and *K*-medians, setting the number of clusters to k = 2. Five variables were used for clustering: (i) total assets plus brokerage, (ii) total assets, (iii) total deposits, (iv) number of branches, and (v) number of clients. The algorithms search iteratively for the best partition using the squared Euclidean distance as the dissimilarity measure. We use only pre-crisis data, from December/2001 through June/2008.

Cluster	Bank
Cluster 1 (Big Banks)	ABN AMRO BB Bradesco CEF HSBC Itau Santander Unibanco
Cluster 2 (Other Banks)	ABC-Brasil Alfa Bancoob Banese Banestes Banif Banpara Banrisul Bansicredi Basa BBM Besc BGN BIC BMG BNB BNP Paribas Bonsucesso Brascan BRB BTMUB BVA Citibank Credit Suisse Cruzeiro do Sul Daycoval DBB BM Deutsche Fibra Ibibank Industrial do Brasil Indusval ING J. Malucelli John Deere JP Morgan Chase Mercantil do Brasil Nossa Caixa SS Pine Prosper Rabobank Rural Safra Schahin SMBC Societe Generale Sofisa UBS Pactual Votorantim WestLB

### Table 2 – Sample

#### Panel A – Distribution of deposits in the sample

Rows [A] to [E] show the proportion of each type of deposit in relation to total deposits in financial institutions of our sample as of December of each year from 2001 to 2009.

	Dec/ 2001	Dec/ 2002	Dec/ 2003	Dec/ 2004	Dec / 2005	Dec/ 2006	Dec/ 2007	Dec/ 2008	Dec/ 2009
[A] Checking account deposits	16%	18%	16%	16%	15%	16%	20%	14%	14%
[B] Savings deposits	34%	32%	31%	30%	26%	26%	27%	23%	26%
[C] Time Deposits	45%	46%	46%	48%	51%	50%	46%	56%	55%
[D] Interbank Deposits	2%	2%	3%	3%	3%	3%	3%	4%	3%
[E] Other Deposits	3%	3%	3%	4%	4%	5%	5%	4%	2%
[F] Total Deposits ([A] + [B] + [C] + [D] +[E])	100%	100%	100%	100%	100%	100%	100%	100%	100%

#### Panel B – Representativeness of sample

Row [A] shows the number of deposit-taking financial institutions as of December of each year from 2001 to 2009, while row [B] shows the number of banks considered in our sample in the same period. In row C, we show the total amount of deposits eligible to receive deposit insurance in the Brazilian Financial System, while in row [D] we show the amount of deposits eligible for deposit insurance for the banks in our sample. In row [E], it is shown the proportion of deposits considered in this study relative to the overall deposits of the Brazilian Financial System.

	Dec/ 2001	Dec/ 2002	Dec/ 2003	Dec/ 2004	Dec / 2005	Dec/ 2006	Dec/ 2007	Dec/ 2008	Dec/ 2009
[A] Number of deposit- taking financial institutions	121	111	110	108	104	104	101	101	100
<b>[B]</b> Number of banks in the sample	74	71	68	65	64	61	60	57	53
<b>[C]</b> Total Deposits (billions of BRL)	313	365	400	470	546	624	740	1,003	1,252
<b>[D]</b> Total Deposits of sample (billions of BRL)	304	357	395	465	535	600	712	986	1,240
[E] Representativeness of sample ([D] / [C])	97.3	97.7	98.9	98.9	97.9	96.1	96.3	98.3	99.1

Table 3 – Summary statisticsMeans and standard deviations (in brackets) are reported for Dec/2001, Jun/2008 and Dec/2009. Bigbanks are defined as in Section II-B.

	Dec	/2001	Jun	/2008	Dec	/2009
	Big banks	Other banks	Big banks	Other banks	Big banks	Other banks
Total agenta (DDI Milliona)	75,488	4,557	271,531	10,985	416,329	11,569
Total assets (DRL Millions)	[45,794]	[6,581]	[132,459]	[15,254]	[208,678]	[16,730]
Total aquity (DDL Millions)	6,335	431	21,606	1.158	35,819	1,270
Total equity (BRL Millions)	[3,024]	[537]	[12,659]	[1.304]	[22,210]	[1,468]
# of denositors (thousands)	10,048	192	16,282	245	23,939	138
# of depositors (mousands)	[7,279]	[648]	[9,989]	[770]	[12,094]	[336]
Uninsured deposits / total	58.6%	87.8%	62.98%	87.0%	61.8%	80.4%
deposits	[13.8%]	[17.3%]	[15.5%]	[19.7%]	[14.1%]	[21.2%]
Fauity ratio	9.7%	14.6%	8.54%	14.56%	9.8%	15.0%
Equity failo	[4.3%]	[8.3%]	[1.97%]	[8.1%]	[5.6%]	[7.9%]
Low quality loops	2.6%	2.5%	2.3%	1.9%	3.2%	2.3%
Low quality loans	[0.7%]	[3.7%]	[0.3%]	[3.5%]	[0.5%]	[2.1%]
Liquidity	34.5%	32.5%	19.2%	27.0%	19.8%	27.8%
Liquidity	[9.5%]	[19.2%]	[9.2%]	[16.4%]	[8.1%]	[18.3%]
Exposure to trade finance	-	-	10.4%	11.0%	8.1%	12.4%
Exposure to trade mance			[4.7%]	[17.0%]	[4.5%]	[16.0%]
Exposure to middle market	-	-	34.8%	34.9%	43.9%	42.6%
Exposure to initiale market			[4.7%]	[27.4%]	[4.9%]	[27.5%]
Foreign funding	10.71%	11.79%	4.4%	9.0%	2.5%	8.5%
roleigh fullding	[6.1%]	[13.7%]	[2.1%]	[10.3%]	[1.8%]	[8.5%]
Reliance on institutional	0.9%	4.7%	3.6%	6.7%	1.6%	8.0%
investors	[0.9%]	[6.0%]	[3.6%]	[9.0%]	[1.6%]	[10.7%]
Deposit concentration	0.00007	1.60	0.00005	0.49	0.00004	0.98
(x1000)	[0.00005]	[8.10]	[0.00006]	[1.35]	[0.00006]	[3.65]
Observations	8	66	8	51	6	46
Exchange Rate (BRL/USD)	2.	.32	2	.34	1.74	

#### Table 4 - Change in deposits, financial crisis and big banks

This table shows the results of the estimation of equation (1) for the change in uninsuredand total deposits using the pooled ordinary least squares (POLS) and (one-step) system GMM estimators. In the GMM regressions, we allow all variables to be only sequentially exogenous, employing suitable lagged values as their instruments, except for the following regressors, which are assumed to be strictly exogenous: time dummies; *Regional economic activity; Big Bank dummy; and Big Bank x Crisis*. Coefficient estimates and autocorrelation/heteroskedasticity-robust t-tests are shown in parentheses. Estimates for the time dummies and MechChange (see Appendix) are omitted. D stands for first difference and L stands for first lag. Variables interacted with the crisis dummy are lagged according to the variable that appears without interaction (for example, *equity ratio x crisis* is the interaction of the crisis dummy with the first lag of the equity ratio, since equity ratio, by itself, is also in first lag). \*, \*\* and \*\*\* indicate statistical significance at 10%, 5% and 1% respectively.

		Uninsured Deposits		Total	Deposits
		(1)	(2)	(3)	(4)
	_	OLS	GMM-Sys	OLS	GMM-Sys
Variables of interest					
Crisis dummy		-0.265	-0.266		-0.216
		(-1.545)	(-1.549)		(-1.254)
Size		0.000	0.000	-0.003	-0.003
		(0.041)	(0.042)	(-0.363)	(-0.361)
Big bank dummy		-0.011	-0.010	-0.012	-0.012
		(-0.337)	(-0.325)	(-0.399)	(-0.384)
Big bank x crisis		$0.448^{***}$	0.447***	0.358***	0.358***
		(3.671)	(3.648)	(3.492)	(3.465)
Control Variables					
Change in uninsured deposits	L	0.038	0.037		
		(0.663)	(0.632)		
Change in total deposits	L			0.048	0.047
				(0.858)	(0.821)
Premium paid on deposits	D	-0.787	-1.100	-0.817	-1.181
		(-0.542)	(-0.784)	(-0.570)	(-0.847)
Equity ratio	L	0.320	0.320	0.284	0.285
		(1.620)	(1.621)	(1.463)	(1.462)
Low quality loans	L	0.099	0.094	-0.236	-0.242
		(0.173)	(0.164)	(-0.423)	(-0.436)
Liquidity	L	0.011	0.011	-0.003	-0.003
		(0.174)	(0.168)	(-0.042)	(-0.050)
Regional economic activity		0.246	0.253	0.050	0.059
		(0.755)	(0.757)	(0.186)	(0.211)
Equity ratio x crisis		-0.675	-0.675	-0.950	-0.950
		(-1.016)	(-1.019)	(-1.457)	(-1.462)
Low quality loans x crisis		-1.702	-1.694	-1.818	-1.809
		(-0.453)	(-0.452)	(-0.497)	(-0.496)
Liquidity x crisis		0.394	0.391	0.381	0.378
		(1.031)	(1.025)	(1.010)	(1.003)
Constant		0.040	0.041	0.154	0.155
		(0.221)	(0.226)	(0.908)	(0.914)
Observations		858	858	858	858
R-squared		0.102		0.085	
F		4.702	4.731	3.494	3.474
F_p			5.68e-08		1.42e-05
hansen		•	56.63		47.00
hansenp		•	1		1
# of cross sections			75		75

#### Table 5 – Exposure to trade finance and middle market loans

This table shows the results of the estimation of equation (1) for the change in uninsured and total deposits using the pooled ordinary least squares (POLS) and (one-step) system GMM estimators. In the GMM regressions, we allow all variables to be only sequentially exogenous, employing suitable lagged values as their instruments, except for the following regressors, which are assumed to be strictly exogenous: time dummies; *Regional economic activity; Big Bank dummy; and Big Bank x Crisis.* Coefficient estimates and autocorrelation/heteroskedasticity-robust t-tests are shown in parentheses. Estimates for the time dummies and MechChange (see Appendix) are omitted. D stands for first difference and L stands for first lag. Variables interacted with the crisis dummy are lagged according to the variable that appears without interaction (for example, *equity ratio x crisis* is the interaction of the crisis dummy with the first lag of the equity ratio, since equity ratio, by itself, is also in first lag). \*, \*\* and \*\*\* indicate statistical significance at 10%, 5% and 1% respectively.

		Uninsured	Deposits	Total	Deposits
		(1)	(2)	(3)	(4)
		OLS	GMM-Sys	OLS	GMM-Sys
Variables of interest			·		· · ·
Crisis dummy		-0.204	-0.377**	-0.132	-0.296*
		(-1.176)	(-2.137)	(-0.772)	(-1.670)
Size		0.001	0.001	-0.002	-0.002
		(0.110)	(0.110)	(-0.183)	(-0.183)
Big bank dummy		-0.007	-0.007	-0.011	-0.011
		(-0.199)	(-0.190)	(-0.328)	(-0.317)
Big bank x crisis		0.482***	0.482***	0.381***	0.381***
-		(3.680)	(3.654)	(3.430)	(3.400)
Control variables					
Exposure to trade finance	L	-0.026	-0.026	-0.038	-0.037
*		(-0.966)	(-0.955)	(-1.427)	(-1.415)
Exposure to middle market	L	0.033	0.033	0.023	0.022
*		(0.956)	(0.953)	(0.670)	(0.664)
Trade finance x crisis		0.158	0.157	0.132	0.131
		(1.316)	(1.316)	(1.101)	(1.098)
Middle market x crisis		-0.145	-0.145	-0.167	-0.167
		(-1.334)	(-1.339)	(-1.556)	(-1.562)
Change in uninsured deposits	L	0.039	0.039		
		(0.671)	(0.668)		
Change in total deposits	L			0.048	0.049
				(0.860)	(0.855)
Premium paid on deposits	D	-0.732	-1.110	-0.764	-1.219
		(-0.506)	(-0.768)	(-0.537)	(-0.853)
Equity ratio	L	0.295	0.294	0.265	0.264
		(1.538)	(1.531)	(1.424)	(1.415)
Low quality loans	L	-0.018	-0.018	-0.392	-0.392
		(-0.032)	(-0.032)	(-0.708)	(-0.710)
Liquidity	L	0.001	0.000	-0.015	-0.016
		(0.020)	(0.007)	(-0.231)	(-0.247)
Regional economic activity		0.200	0.202	0.028	0.028
		(0.615)	(0.609)	(0.102)	(0.100)
Equity ratio x crisis		-0.296	-0.295	-0.591	-0.590
		(-0.427)	(-0.427)	(-0.864)	(-0.865)
Low quality loans x crisis		-0.808	-0.816	-1.390	-1.400
		(-0.218)	(-0.221)	(-0.380)	(-0.383)
Liquidity x crisis		0.546	0.544	0.503	0.501
		(1.462)	(1.459)	(1.345)	(1.342)
Constant		-0.133	0.041	-0.015	0.150
		(-0.751)	(0.215)	(-0.089)	(0.824)
Observations		858	858	858	858
R-squared		0.108		0.091	
F		4.723	4.700	3.216	3.203
F-p			2.53e-08		2.54e-05
Hansen			44.27		42.62
Hansen-p		•	1		1
# of cross sections			75		75

#### Table 6 – Exposure to foreign funding

This table shows the results of the estimation of equation (1) for the change in uninsured and total deposits using the pooled ordinary least squares (POLS) and (one-step) system GMM estimators. In the GMM regressions, we allow all variables to be only sequentially exogenous, employing suitable lagged values as their instruments, except for the following regressors, which are assumed to be strictly exogenous: time dummies; *Regional economic activity; Big Bank dummy; and Big Bank x Crisis.* Coefficient estimates and autocorrelation/heteroskedasticity-robust t-tests are shown in parentheses. Estimates for the time dummies and *mechanical change in deposits* (see Appendix) are omitted. D stands for first difference and L stands for first lag. Variables interacted with the crisis dummy are lagged according to the variable that appears without interaction (for example, *equity ratio x crisis* is the interaction of the crisis dummy with the first lag of the equity ratio, since equity ratio, by itself, is also in first lag). \*, \*\* and \*\*\* indicate statistical significance at 10%, 5% and 1% respectively.

		Uninsured	Deposits	Total	Deposits
		(1)	(2)	(3)	(4)
		OLS	GMM-Sys	OLS	GMM-Sys
Variables of interest			•		
Crisis dummy		-0.328*	-0.329*		-0.309*
		(-1.982)	(-1.991)		(-1.949)
Size		0.000	0.000	-0.003	-0.004
		(0.034)	(0.037)	(-0.355)	(-0.388)
Big bank dummy		-0.013	-0.013	-0.014	-0.013
· ·		(-0.383)	(-0.380)	(-0.428)	(-0.402)
Big bank x crisis		0.474***	0.474***	0.380***	0.380***
C		(3.861)	(3.842)	(3.684)	(3.653)
Control variables					
Foreign funding	L	-0.081	-0.082	-0.066	-0.063
		(-0.635)	(-0.642)	(-0.513)	(-0.484)
Foreign funding x crisis		0.811	0.807	0.685	0.666
6 6		(1.524)	(1.520)	(1.275)	(1.244)
Change in uninsured deposits	L	0.037	0.037	, ,	
		(0.649)	(0.649)		
Change in total deposits	L			0.047	0.039
				(0.840)	(0.740)
Premium paid on deposits	D	-0.777	-1.007	-0.808	-1.199
		(-0.534)	(-0.704)	(-0.563)	(-0.840)
Equity ratio	L	0.318	0.318	0.283	0.277
		(1.580)	(1.575)	(1.433)	(1.404)
Low quality loans	L	0.029	0.026	-0.296	-0.339
		(0.048)	(0.044)	(-0.512)	(-0.588)
Liquidity	L	0.007	0.006	-0.006	-0.010
		(0.105)	(0.092)	(-0.095)	(-0.149)
Regional economic activity		0.171	0.184	-0.013	0.056
		(0.503)	(0.532)	(-0.046)	(0.193)
Equity ratio x crisis		-0.832	-0.830	-1.083*	-1.076*
		(-1.326)	(-1.325)	(-1.739)	(-1.726)
Low quality loans x crisis		-0.789	-0.790	-1.044	-0.992
		(-0.213)	(-0.214)	(-0.292)	(-0.278)
Liquidity x crisis		0.410	0.410	0.395	0.394
		(1.130)	(1.129)	(1.096)	(1.089)
Constant		0.060	0.059	0.171	0.209
		(0.328)	(0.328)	(1.002)	(1.258)
Observations		858	858	858	858
R-squared		0.108		0.091	
F		4.723	4.700	3.216	3.203
F_p			2.53e-08		2.54e-05
hansen			44.27		42.62
hansenp		•	1	•	1
# of cross sections			75		75

#### **Table 7 – Deposit concentration**

This table shows the results of the estimation of equation (1) for the change in uninsuredand total deposits using the pooled ordinary least squares (POLS) and (one-step) system GMM estimators. In the GMM regressions, we allow all variables to be only sequentially exogenous, employing suitable lagged values as their instruments, except for the following regressors, which are assumed to be strictly exogenous: time dummies; *Regional economic activity; Big Bank dummy; and Big Bank x Crisis*. Coefficient estimates and autocorrelation/heteroskedasticity-robust t-tests are shown in parentheses. Estimates for the time dummies and *mechanical change in deposits* (see Appendix) are omitted. D stands for first difference and L stands for first lag. Variables interacted with the crisis dummy are lagged according to the variable that appears without interaction (for example, *equity ratio x crisis* is the interaction of the crisis dummy with the first lag of the equity ratio, since equity ratio, by itself, is also in first lag). \*, \*\* and \*\*\* indicate statistical significance at 10%, 5% and 1% respectively.

		Uninsured	Deposits	Total	Deposits
		(1)	(2)	(3)	(4)
		OLS	GMM-Sys	OLS	GMM-Sys
Variables of interest					
Crisis dummy			-0.426		-0.453*
			(-1.627)		(-1.694)
Size		-0.006	-0.006	-0.010	-0.010
		(-0.598)	(-0.618)	(-0.968)	(-0.970)
Big bank dummy		-0.033	-0.033	-0.031	-0.031
		(-0.850)	(-0.841)	(-0.858)	(-0.851)
Big bank x crisis		0.396***	0.393***	0.261**	0.259**
		(2.673)	(2.648)	(2.070)	(2.052)
Control variables					
Deposit concentration	L	-0.007	-0.007	-0.006	-0.006
		(-1.254)	(-1.269)	(-1.234)	(-1.244)
Deposit concentration x crisis		-0.012	-0.013	-0.021	-0.022
-		(-0.594)	(-0.625)	(-1.062)	(-1.081)
Change in uninsured deposits	L	0.040	0.038		
C 1		(0.686)	(0.657)		
Change in total deposits	L			0.049	0.050
				(0.866)	(0.866)
Premium paid on deposits	D	-0.832	-1.373	-0.868	-1.358
		(-0.575)	(-0.954)	(-0.608)	(-0.952)
Equity ratio	L	0.352*	0.348*	0.311	0.309
		(1.686)	(1.666)	(1.512)	(1.502)
Low quality loans	L	-0.232	-0.257	-0.546	-0.550
		(-0.373)	(-0.414)	(-0.937)	(-0.945)
Liquidity	L	-0.004	-0.007	-0.018	-0.019
		(-0.062)	(-0.106)	(-0.264)	(-0.288)
Regional economic activity		0.259	0.299	0.087	0.099
-		(0.769)	(0.880)	(0.318)	(0.355)
Equity ratio x crisis		-0.413	-0.399	-0.530	-0.523
		(-0.582)	(-0.563)	(-0.749)	(-0.742)
Low quality loans x crisis		-3.019	-3.037	-3.929	-3.947
		(-0.735)	(-0.740)	(-0.983)	(-0.989)
Liquidity x crisis		0.354	0.351	0.314	0.312
		(0.920)	(0.915)	(0.842)	(0.838)
Constant		-0.318	0.100	-0.250	0.200
		(-0.998)	(0.541)	(-0.796)	(1.138)
Observations		858	858	858	858
R-squared		0.105		0.088	
F		5.294	5.475	4.035	3.975
F-p			2.35e-09		1.07e-06
Hansen			47.67		55.31
Hansen-p			1		1
# of cross sections			75		75

#### Table 8 – Reliance on institutional investors for funding

This table shows the results of the estimation of equation (1) for the change in uninsured and total deposits using the pooled ordinary least squares (POLS) and (one-step) system GMM estimators. In the GMM regressions, we allow all variables to be only sequentially exogenous, employing suitable lagged values as their instruments, except for the following regressors, which are assumed to be strictly exogenous: time dummies; *Regional economic activity; Big Bank dummy; and Big Bank x Crisis.* Coefficient estimates and autocorrelation/heteroskedasticity-robust t-tests are shown in parentheses. Estimates for the time dummies and *mechanical change in deposits* (see Appendix) are omitted. D stands for first difference and L stands for first lag. Variables interacted with the crisis dummy are lagged according to the variable that appears without interaction (for example, *equity ratio x crisis* is the interaction of the crisis dummy with the first lag of the equity ratio, since equity ratio, by itself, is also in first lag). \*, \*\* and \*\*\* indicate statistical significance at 10%, 5% and 1% respectively.

		Uninsured	Deposits	Total	Deposits
		(1)	(2)	(3)	(4)
		OLS	GMM-Sys	OLS	GMM-Sys
Variables of interest					
Crisis dummy		-0.352**	-0.353**	-0.298**	-0.300**
		(-2.502)	(-2.511)	(-2.141)	(-2.151)
Size		-0.006	-0.006	-0.009	-0.009
		(-0.608)	(-0.607)	(-0.959)	(-0.958)
Big bank dummy		-0.000	0.000	-0.002	-0.002
<i>c i</i>		(-0.007)	(0.004)	(-0.071)	(-0.059)
Big bank x crisis		0.381***	0.381***	0.292***	0.292***
6		(3.594)	(3.560)	(3.361)	(3.325)
Control variables			× /		
Reliance on institutional investors	L	-0.193	-0.196	-0.145	-0.148
		(-1.215)	(-1.229)	(-0.888)	(-0.904)
Reliance on inst. investors x		-2.884***	-2.876***	-2.867***	-2.859***
C11818		(6.277)	(6.283)	(6252)	(6 250)
Change in uninguned demosite	т	(-0.277)	(-0.203)	(-0.232)	(-0.239)
Change in uninsured deposits	L	(0.044)	(0.043)		
Change in total deposits	т	(0.704)	(0.708)	0.055	0.055
Change in total deposits	L			0.033	(0.033)
Decesions and an demosite	D	0.649	1.054	(0.970)	(0.972)
Premium paid on deposits	D	-0.648	-1.054	-0.070	-1.114
<b>F</b> '/ /:	т	(-0.447)	(-0.741)	(-0.4/4)	(-0.792)
Equity ratio	L	0.336	0.335	0.289	0.288
T 11. 1		(1.657)	(1.649)	(1.466)	(1.457)
Low quality loans	L	0.269	0.267	-0.090	-0.093
<b>T</b> • • • • •		(0.469)	(0.466)	(-0.162)	(-0.166)
Liquidity	L	-0.005	-0.006	-0.015	-0.017
		(-0.065)	(-0.083)	(-0.219)	(-0.239)
Regional economic activity		0.392	0.398	0.193	0.202
		(1.286)	(1.278)	(0.769)	(0.785)
Equity ratio x crisis		0.945	0.943	0.648	0.647
		(1.458)	(1.459)	(1.023)	(1.023)
Low quality loans x crisis		0.105	0.105	-0.031	-0.030
		(0.053)	(0.053)	(-0.016)	(-0.015)
Liquidity x crisis		0.467*	0.465*	0.449*	0.448*
		(1.813)	(1.809)	(1.785)	(1.780)
Constant		0.126	0.127	0.230	0.231
		(0.703)	(0.713)	(1.345)	(1.353)
Observations		852	852	852	852
R-squared		0.132		0.115	
F		13.17	13.26	14.35	14.52
F-p			0		0
Hansen			50.42		44.62
Hansen-p			1		1
# of cross sections			73		73

#### Table 9 – The effect of government-owned banks

This table shows the results of the estimation of equation (1) for the change in uninsured and total deposits using the pooled ordinary least squares (POLS) and (one-step) system GMM estimators. In the GMM regressions, we allow all variables to be only sequentially exogenous, employing suitable lagged values as their instruments, except for the following regressors, which are assumed to be strictly exogenous: time dummies; *Regional economic activity; Big Bank dummy; and Big Bank x Crisis.* Coefficient estimates and autocorrelation/heteroskedasticity-robust t-tests are shown in parentheses. Estimates for the time dummies and MechChange (see Appendix) are omitted. D stands for first difference and L stands for first lag. Variables interacted with the crisis dummy are lagged according to the variable that appears without interaction (for example, *equity ratio x crisis* is the interaction of the crisis dummy with the first lag of the equity ratio, since equity ratio, by itself, is also in first lag). \*, \*\* and \*\*\* indicate statistical significance at 10%, 5% and 1% respectively.

		Uninsured Deposits		Total	Deposits
		(1)	(2)	(3)	(4)
		OLS	GMM-Sys	OLS	GMM-Sys
Variables of interest					
Crisis dummy		-0.268	-0.268	-0.217	-0.218
2		(-1.588)	(-1.590)	(-1.281)	(-1.284)
Size		0.000	0.000	-0.003	-0.003
		(0.021)	(0.021)	(-0.360)	(-0.358)
Big bank dummy		-0.010	-0.010	-0.011	-0.011
0		(-0.322)	(-0.312)	(-0.364)	(-0.351)
Big bank x crisis		0.417***	0.417***	0.328***	0.328***
e		(3.581)	(3.568)	(3.236)	(3.223)
Control variables				· · · ·	
Government-ownedbank		0.001	0.001	-0.023	-0.023
dummy					
		(0.041)	(0.052)	(-1.206)	(-1.193)
Government-owned bank x		0.166	0.165	0.180*	0.180*
crisis		(1.50)		(1.0.62)	(1.0.50)
		(1.652)	(1.656)	(1.962)	(1.966)
Change in uninsured deposits	L	0.037	0.036		
	-	(0.637)	(0.612)	0.047	0.046
Change in total deposits	L			0.047	0.046
<b>N</b>	P	0.5.0	1.01.4	(0.839)	(0.802)
Premium paid on deposits	D	-0.763	-1.014	-0.768	-1.055
	-	(-0.528)	(-0.724)	(-0.539)	(-0.757)
Equity ratio	L	0.319	0.319	0.265	0.266
	-	(1.581)	(1.583)	(1.362)	(1.364)
Low quality loans	L	0.086	0.080	-0.079	-0.086
	-	(0.139)	(0.129)	(-0.129)	(-0.142)
Liquidity	L	0.010	0.009	0.019	0.019
		(0.131)	(0.124)	(0.263)	(0.256)
Regional economic activity		0.321	0.324	0.161	0.165
		(0.994)	(0.980)	(0.604)	(0.606)
Equity ratio x crisis		-0.460	-0.461	-0.729	-0.730
		(-0.654)	(-0.657)	(-1.055)	(-1.058)
Low quality loans x crisis		-3.148	-3.139	-3.340	-3.328
		(-0.826)	(-0.824)	(-0.898)	(-0.895)
Liquidity x crisis		0.247	0.246	0.221	0.219
_		(0.601)	(0.599)	(0.541)	(0.537)
Constant		0.036	0.037	0.139	0.140
		(0.197)	(0.203)	(0.813)	(0.821)
Observations		858	858	858	858
R-squared		0.104		0.087	
F		4.570	4.540	3.617	3.567
F-p			7.72e-08	•	6.54e-06
Hansen			54.61	•	54.39
Hansen-p		•	1	•	1
# of cross sections			75		75

Table 10 – The effect on big government-owned banks (excluding privately-owned banks from the sample)

This table shows the results of the estimation of equation (1) for the change in uninsured and total deposits using the pooled ordinary least squares (POLS) and (one-step) system GMM estimators. In the GMM regressions, we allow all variables to be only sequentially exogenous, employing suitable lagged values as their instruments, except for the following regressors, which are assumed to be strictly exogenous: time dummies; *Regional economic activity; Big Government-Owned Bank; and Big Government-Owned Bank x Crisis.* Coefficient estimates and autocorrelation/heteroskedasticity-robust t-tests are shown in parentheses. Estimates for the time dummies and MechChange (see Appendix) are omitted. D stands for first difference and L stands for first lag. Variables interacted with the crisis dummy are lagged according to the variable that appears without interaction (for example, *equity ratio x crisis* is the interaction of the crisis dummy with the first lag of the equity ratio, since equity ratio, by itself, is also in first lag). \*, \*\* and \*\*\* indicate statistical significance at 10%, 5% and 1% respectively.

		Uninsured	l Deposits	Total	Deposits
		(1)	(2)	(3)	(4)
	_	OLS	GMM-Sys	OLS	GMM-Sys
Variables of interest	_				
Crisis dummy		-0.278	-0.299*	-0.226	-0.226
		(-1.586)	(-1.806)	(-1.281)	(-1.292)
Size		0.001	0.001	-0.003	-0.003
		(0.122)	(0.105)	(-0.290)	(-0.289)
Big government-owned bank dummy		-0.024	-0.023	-0.019	-0.019
		(-0.724)	(-0.713)	(-0.620)	(-0.602)
Big gov. bank x crisis		0.455**	0.457**	0.342**	0.342**
		(2.314)	(2.286)	(2.136)	(2.101)
Control Variables					
Change in uninsured deposits	L	0.045	0.040		
		(0.733)	(0.674)		
Change in total deposits	L			0.056	0.056
				(0.948)	(0.921)
Premium paid on deposits	D	-0.832	-1.352	-0.825	-1.272
		(-0.552)	(-0.913)	(-0.553)	(-0.866)
Equity ratio	L	0.321	0.318	0.285	0.284
		(1.588)	(1.571)	(1.432)	(1.428)
Low quality loans	L	0.076	0.053	-0.252	-0.256
		(0.129)	(0.091)	(-0.442)	(-0.452)
Liquidity	L	0.009	0.007	-0.004	-0.005
		(0.135)	(0.104)	(-0.058)	(-0.073)
Regional economic activity		0.252	0.283	0.049	0.052
		(0.770)	(0.854)	(0.181)	(0.189)
Equity ratio x crisis		-0.699	-0.698	-0.972	-0.975
		(-1.039)	(-1.043)	(-1.474)	(-1.485)
Low quality loans x crisis		-1.701	-1.684	-1.786	-1.782
		(-0.445)	(-0.443)	(-0.480)	(-0.480)
Liquidity x crisis		0.402	0.396	0.394	0.391
		(1.021)	(1.009)	(1.016)	(1.010)
Constant		0.040	0.062	0.153	0.156
		(0.218)	(0.343)	(0.875)	(0.891)
Observations		776	776	776	776
R-squared		0.109		0.095	
F		6.438	5.893	5.506	5.502
F-p		•	2.87e-09	•	1.08e-08
Hansen		•	47.68	•	44.31
Hansen-p			1		1
# of cross sections			67		67

Table 11 – The effect on big privately-owned banks (excluding government-owned banks from the sample)

This table shows the results of the estimation of equation (1) for the change in uninsured and total deposits using the pooled ordinary least squares (POLS) and (one-step) system GMM estimators. In the GMM regressions, we allow all variables to be only sequentially exogenous, employing suitable lagged values as their instruments, except for the following regressors, which are assumed to be strictly exogenous: time dummies; *Regional economic activity; Big Government-Owned Bank; and Big Government-Owned Bank x Crisis*. Coefficient estimates and autocorrelation/heteroskedasticity-robust t-tests are shown in parentheses. Estimates for the time dummies and MechChange (see Appendix) are omitted. D stands for first difference and L stands for first lag. Variables interacted with the crisis dummy are lagged according to the variable that appears without interaction (for example, *equity ratio x crisis* is the interaction of the crisis dummy with the first lag of the equity ratio, since equity ratio, by itself, is also in first lag). \*, \*\* and \*\*\* indicate statistical significance at 10%, 5% and 1% respectively.

		Uninsured	Deposits	Total	Deposits
		(1)	(2)	(3)	(4)
		OLS	GMM-Sys	OLS	GMM-Sys
Variables of interest					
Crisis dummy		-0.281	-0.283	-0.230	-0.232
		(-1.592)	(-1.602)	(-1.294)	(-1.305)
Size		0.000	0.000	-0.003	-0.003
		(0.034)	(0.037)	(-0.372)	(-0.368)
Big private bank dummy		-0.007	-0.006	-0.010	-0.010
		(-0.204)	(-0.191)	(-0.308)	(-0.295)
Big private bank x crisis		0.451***	0.450***	0.380***	0.379***
		(3.885)	(3.882)	(3.800)	(3.797)
Control Variables					
Change in uninsured deposits	L	0.039	0.038		
		(0.671)	(0.640)		
Change in total deposits	L			0.051	0.049
				(0.892)	(0.855)
Premium paid on deposits	D	-0.786	-1.193	-0.800	-1.230
		(-0.539)	(-0.841)	(-0.555)	(-0.872)
Equity ratio	L	0.314	0.314	0.281	0.281
		(1.581)	(1.579)	(1.437)	(1.435)
Low quality loans	L	0.111	0.105	-0.232	-0.240
		(0.193)	(0.183)	(-0.414)	(-0.429)
Liquidity	L	0.014	0.013	-0.002	-0.002
		(0.204)	(0.192)	(-0.024)	(-0.036)
Regional economic activity		0.252	0.266	0.053	0.066
		(0.770)	(0.797)	(0.196)	(0.240)
Equity ratio x crisis		-0.687	-0.687	-0.964	-0.963
		(-1.034)	(-1.038)	(-1.478)	(-1.483)
Low quality loans x crisis		-1.602	-1.592	-1.701	-1.691
		(-0.420)	(-0.419)	(-0.459)	(-0.457)
Liquidity x crisis		0.455	0.453	0.436	0.434
		(1.127)	(1.123)	(1.092)	(1.088)
Constant		0.039	0.039	0.154	0.155
		(0.215)	(0.217)	(0.901)	(0.905)
		020	000	020	000
Observations		829	829	829	829
K-squared		0.101	5.250	0.086	4 220
F		5.137	5.250	4.230	4.228
F-р			1.06e-08		7.30e-07
Hansen			50.19		54.36
Hansen-p			1		1
# of cross sections			72		72

S
12
Ę
12
1
H
2
-
2
60
S
E
.2
Ð
5
×
3
- <b>ວ</b>
~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~
Ś
Ľ,
Ś
e .
1
.=
3
n
0
:=
3
1
1
2
_
2
-
പ
Ť.
9
<u> </u>
_

4), non-financial firms (5 to 8) and individuals (9 to 12) using (one-step) system GMM estimators, in which we allow all variables to be only sequentially exogenous, employing suitable lagged values as their instruments, except for the following regressors, which are assumed to be strictly exogenous: time dumnies; *Regional* This table shows the results of the estimation of equation (1) for the change in the amount of certificates of deposits held by institutional investors (specifications 1 to economic activity; Big Bank dummy; and Big Bank x Crisis. Coefficient estimates and autocorrelation/heteroskedasticity-robust t-tests are shown in parentheses. Estimates for the following variables are omitted: the first lag of the dependent variable, time dummies, premium paid on deposits (lagged difference) and mechanical change in deposits (see Appendix) and the intercept. D stands for first difference and L stands for first lag. Variables interacted with the crisis dummy are lagged according to the variable that appears without interaction (for example, equity ratio x crisis is the interaction of the crisis dummy with the first lag of the equity ratio, since equity ratio, by itself, is also in first lag). \*, \*\* and \*\*\* indicate statistical significance at 10%, 5% and 1% respectively.

		Institution	d investors			Non-finan	cial firms			Indivi	duals	
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
Variables of interest												
Crisis dummy	0.189	-0.518	0.132	-0.001	0.189	-0.134	-0.117	-0.227	$0.246^{*}$	0.296	0.223	$0.242^{*}$
	(0.439)	(-1.366)	(0.285)	(-0.003)	(0.439)	(-0.625)	(-0.620)	(-1.444)	(1.734)	(1.546)	(1.599)	(1.703)
Size	-0.008	-0.007	-0.010	-0.019	-0.020	-0.019	-0.020	-0.019	0.001	-0.001	0.001	0.002
	(-0.537)	(-0.442)	(-0.653)	(-1.108)	(-1.594)	(-1.461)	(-1.575)	(-1.449)	(0.080)	(-0.047)	(0.069)	(0.168)
Big bank dummy	0.086	0.091	0.096	$0.101^{*}$	0.064	0.060	0.066	0.070	0.055	0.061	0.055	0.056
	(1.545)	(1.486)	(1.612)	(1.675)	(1.459)	(1.339)	(1.472)	(1.554)	(1.189)	(1.320)	(1.121)	(1.194)
Big bank x crisis	$0.722^{**}$	$0.978^{***}$	$0.750^{**}$	0.577*	$0.451^{***}$	$0.449^{***}$	$0.444^{***}$	$0.403^{***}$	0.163	0.116	0.172	0.154
	(2.094)	(2.854)	(2.130)	(1.896)	(3.055)	(2.830)	(2.964)	(2.736)	(1.361)	(0.888)	(1.426)	(1.274)
<b>Control variables</b>												
Equity ratio 1	, 1.128***	$1.084^{***}$	$1.116^{***}$	$1.264^{***}$	-0.107	-0.099	-0.109	-0.197	-0.010	-0.021	-0.013	-0.057
	(4.082)	(3.819)	(3.985)	(4.076)	(-0.436)	(-0.411)	(-0.441)	(-0.810)	(-0.066)	(-0.128)	(-0.080)	(-0.369)
Low quality loans 1	, -0.990	-1.348	-0.885	-0.767	0.274	0.213	0.307	0.138	-0.070	-0.019	-0.089	-0.124
	(-1.066)	(-1.474)	(606.0-)	(-0.761)	(0.429)	(0.336)	(0.471)	(0.215)	(-0.173)	(-0.048)	(-0.224)	(-0.325)
Liquidity 1	0.018	-0.005	0.026	-0.031	0.108	0.107	0.110	0.126	0.018	0.021	0.016	0.030
	(0.140)	(-0.037)	(0.203)	(-0.231)	(1.354)	(1.313)	(1.441)	(1.469)	(0.357)	(0.410)	(0.318)	(0.577)
Regional economic activity	-0.499	-0.688	-0.553	-0.120	0.420	0.411	0.432	0.557*	0.204	0.158	0.180	0.236
5	(-0.712)	(-0.952)	(-0.770)	(-0.175)	(1.407)	(1.281)	(1.409)	(1.888)	(1.069)	(0.793)	(0.949)	(1.233)
Equity ratio x crisis	-2.190	-0.939	-2.398	1.097	-0.643	-0.588	-0.603	0.408	-0.953	-0.620	-1.006	-0.770
	(-1.419)	(-0.616)	(-1.629)	(0.674)	(-0.836)	(-0.718)	(-0.793)	(0.603)	(-1.449)	(-0.992)	(-1.482)	(-1.295)
Low quality loans x crisis	-5.408	2.345	-4.386	-1.650	0.055	-0.130	-0.191	1.318	-1.023	-2.342	-0.696	-0.821
	(-0.630)	(0.294)	(-0.489)	(-0.333)	(0.020)	(-0.046)	(-0.069)	(0.465)	(-0.548)	(-0.848)	(-0.359)	(-0.413)
Liquidity x crisis	-0.790	0.033	-0.779	-0.625	0.208	0.221	0.202	0.220	0.012	0.024	0.017	0.005
	(-0.852)	(0.043)	(-0.843)	(-0.795)	(0.557)	(0.555)	(0.549)	(0.683)	(0.061)	(0.089)	(0.087)	(0.025)
<b>Other</b> variables												
Exposure to trade		-0.110**				-0.021				0.014		
		(-2.195)				(-0.606)				(0.528)		
Exposure to middle		0.013				-0.023				0.010		

							0.020	0.151)	.287		.781)	0.163		(1.140)	-0.401		(-0.750)	949 949	8.773 8.857	0 0	8.95 44.19	1 1	74 74
	(0.427)	-0.018	(-0.176)	-0.239***		(-2.802)	Ŷ	)-)	0		0)							949	10.87 8	0	38.78 3	1	74
																		949	9.267	0	42.12	1	74
												0.278		(1.076)	-2.059***		(-3.966)	949	9.256	0	50.22	1	74
							0.035	(0.208)	-0.218		(-0.434)							949	8.361	0	50.36	1	74
	(-0.603)	0.016	(0.144)	-0.046		(-0.452)												949	8.134	0	47.99	1	74
																		949	7.697	0	52.27	1	74
												-0.660*		(-1.885)	-5.743***		(-5.554)	949	11.15	0	43.70	1	74
							0.098	(0.491)	0.856		(0.798)							949	3.971	9.72e-07	41.37	1	74
	(0.205)	$0.854^{***}$	(3.199)	-0.197		(-0.818)												949	5.986	1.76e-10	47.95	1	74
																		949	4.491	1.66e-07	40.96	1	74
market	;	Trade finance x	CISID	Middle market x	crisis		Foreign funding		Foreign Funding x	crisis		Reliance on instit.	investors		Instit. Investors x	crisis		Observations	F	F-p	Hansen	hansenp	# of cross sections

#### Table 13 – Change in deposits, post-financial crisis and big banks

This table shows the results of the estimation of equation (1) for the change in uninsured and total deposits using the pooled ordinary least squares (POLS) and (one-step) system GMM estimators. In the GMM regressions, we allow all variables to be only sequentially exogenous, employing suitable lagged values as their instruments, except for the following regressors, which are assumed to be strictly exogenous: time dummies; *Regional economic activity; Big Bank dummy; Big Bank x Crisis and Big Bank x Post-Crisis*. Coefficient estimates and autocorrelation/heteroskedasticity-robust t-tests are shown in parentheses. Estimates for the time dummies and *mechanical change in deposits* (see Appendix) are omitted. D stands for first difference and L stands for first lag. Variables interacted with the crisis dummy and the post-crisis dummy are lagged according to the variable that appears without interaction (for example, *equity ratio x crisis* is the interaction of the crisis dummy with the first lag of the equity ratio, since equity ratio, by itself, is also in first lag). \*, \*\* and \*\*\* indicate statistical significance at 10%, 5% and 1% respectively.

		Uninsured	Deposits	Total	Deposits
		(1)	(2)	(3)	(4)
		OLS	GMM-Sys	OLS	GMM-Sys
Variables of interest					
Crisis dummy		-0.191	-0.398***	-0.142	-0.324**
-		(-1.438)	(-2.989)	(-1.086)	(-2.429)
Post-Crisis dummy		0.119	-0.088	0.054	-0.128
-		(1.071)	(-0.878)	(0.509)	(-1.398)
Size		-0.009	-0.009	-0.012	-0.012
		(-1.016)	(-1.014)	(-1.366)	(-1.364)
Big bank dummy		0.024	0.024	0.019	0.019
		(0.729)	(0.735)	(0.570)	(0.577)
Big bank x crisis		0.366***	0.366***	0.280***	0.281***
-		(3.485)	(3.454)	(3.236)	(3.203)
Big bank x post-crisis		-0.150*	-0.149*	-0.134**	-0.133*
		(-1.876)	(-1.859)	(-1.994)	(-1.971)
Control variables					
Rel.on institutional investors	L	-0.276	-0.279*	-0.282*	-0.285*
		(-1.657)	(-1.673)	(-1.743)	(-1.761)
Rel. on inst. investors x crisis		-2.811***	-2.802***	-2.736***	-2.728***
		(-5.990)	(-5.993)	(-5.854)	(-5.856)
Rel. on inst. investors x post-	L	0.976**	0.981**	1.474***	1.479***
crisis					
		(2.526)	(2.540)	(3.506)	(3.522)
Premium paid on deposits	D	-0.646	-1.052	-0.702	-1.142
		(-0.442)	(-0.734)	(-0.488)	(-0.806)
Equity ratio	L	0.247	0.246	0.185	0.183
		(1.163)	(1.154)	(0.922)	(0.912)
Low quality loans	L	-0.087	-0.091	-0.470	-0.474
		(-0.151)	(-0.157)	(-0.832)	(-0.839)
Liquidity	L	0.011	0.010	-0.016	-0.018
		(0.159)	(0.136)	(-0.241)	(-0.267)
Regional economic activity		0.376	0.382	0.162	0.172
		(1.319)	(1.314)	(0.688)	(0.711)
Liquidity x crisis		0.451*	0.450*	0.449*	0.448*
		(1.702)	(1.699)	(1.719)	(1.715)
Constant		0.026	0.234	0.136	0.318*
		(0.163)	(1.310)	(0.892)	(1.918)
Lagged dependent variable	L	Yes	Yes	Yes	Yes
Other interactions controls x		Yes	Yes	Yes	Yes
crisis					
Other interactions controls x		Yes	Yes	Yes	Yes
post-crisis					
Observations		852	852	852	852
R-squared		0.148		0.137	
F		13.10	13.17	14.14	14.20
F-p			0		0
Hansen			47.94		41.68

Hansen-p	1	1
# of cross sections	73	73

This table shows the result: 4), non-financial firms (5 to employing suitable lagged <i>economic activity; Big Bai</i> shown in parentheses. Esti difference) and <i>mechanica</i> , crisis dummy are lagged ac relisis dummy are lagged ac	s of the est o 8) and in values as <i>nk dummy;</i> mates for <i>l change in</i> cording to s equity rat	imation of equividuals (9 tuble their instrume Big Bank $x$ (1) the following the following the variable the	attion (1) fc attion (1) fc ang, except <i>Crisis and</i> J variables a Appendix) tat appears v also in first	or the chang (one-step) s for the fol Big Bank x re omitted: and the int without inte vithout inte	e in the am ystem GM lowing reg <i>Post-Crisi</i> the first la ercept. D s raction (for and *** in	Nount of cer M estimato Tressors, wh s. Coefficie g of the de tands for fu tands for fu tands catis	tificates of rs, in which rich are ass int estimate spendent va rest differen quity ratio stical signif	deposits he deposits he $1$ we allow $z$ sumed to be seared autoor with autoor with the transformed to the $x$ crisis is the $x$ crisis is the formed of th	ld by institu all variables e strictly ex correlation/I e dummies, ands for fire ne interactic %, 5% and	trional inves to be only s cogenous: tri neteroskedas premium ps tr lag. Varia on of the cris on of the cris	tors (specifi sequentially me dummie sticity-robus sticity-robus aid on depo bles interact is dummy v vely.	cations 1 to exogenous, s; <i>Regional</i> t t-tests are sits (lagged ed with the vith the first
	10	Institutiona	l investors	Ŵ	(2)	Non-finan	cial firms	(8)	(0)	Individ	duals	(61)
Variahles of interest	(1)	(7)	(c)	(+)	(c)	(0)	(1)	(0)		(11)	(11)	(17)
Crisis dummy	0.119	-0.594	0.044	-0.062	-0.329*	-0.327	-0.309	-0.362**	-0.015	0.033	-0.041	-0.020
	(0.288)	(-1.633)	(0.098)	(-0.170)	(-1.774)	(-1.553)	(-1.641)	(-2.190)	(-0.095)	(0.162)	(-0.274)	(-0.131)
Post-Crisis dummy	-0.108	-0.073	-0.142	-0.128	-0.319**	-0.288**	-0.270**	-0.229*	-0.157**	-0.131	$-0.180^{**}$	-0.032
	(-0.352)	(-0.238)	(-0.478)	(-0.423)	(-2.491)	(-2.003)	(-2.072)	(-1.871)	(-2.095)	(-1.441)	(-2.076)	(-0.385)
Size	-0.013	-0.013	-0.014	-0.024	-0.023*	-0.022*	-0.023*	-0.022	-0.000	-0.001	-0.000	0.001
	(-0.912)	(-0.835)	(-0.917)	(-1.430)	(-1.801)	(-1.685)	(-1.783)	(-1.644)	(-0.025)	(-0.089)	(-0.004)	(0.054)
Big bank dummy	$0.144^{**}$	$0.159^{**}$	$0.147^{**}$	$0.156^{**}$	$0.085^{*}$	$0.082^{*}$	$0.086^{*}$	0.090*	0.070	0.073	0.069	0.070
	(2.456)	(2.519)	(2.370)	(2.526)	(1.786)	(1.722)	(1.817)	(1.863)	(1.477)	(1.511)	(1.369)	(1.452)
Big bank x crisis	$0.680^{*}$	$0.939^{***}$	$0.711^{*}$	$0.536^{*}$	$0.438^{***}$	$0.433^{***}$	$0.431^{***}$	$0.390^{***}$	0.151	0.105	0.161	0.144
	(1.937)	(2.736)	(1.984)	(1.729)	(3.024)	(2.766)	(2.921)	(2.698)	(1.243)	(0.793)	(1.311)	(1.167)
Big bank x post-crisis	-0.428*	-0.491**	-0.400*	-0.412*	-0.126	-0.149	-0.164	-0.132	-0.128	-0.137*	-0.112	-0.128
	(-1.937)	(-2.065)	(-1.704)	(-1.873)	(-1.104)	(-1.225)	(-1.430)	(-1.141)	(-1.651)	(-1.669)	(-1.394)	(-1.649)
<b>Control</b> variables												
Equity ratio L	0.934***	$0.900^{**}$	$0.928^{***}$	$1.070^{***}$	-0.241	-0.232	-0.243	-0.324	0.048	0.039	0.045	0.018
	(2.834)	(2.626)	(2.789)	(2.950)	(-0.927)	(-0.897)	(-0.933)	(-1.263)	(0.284)	(0.228)	(0.263)	(0.106)
Low quality loans L	-1.329	-1.693*	-1.323	-1.127	-0.036	-0.064	0.015	-0.136	-0.156	-0.119	-0.216	-0.211
	(-1.442)	(-1.854)	(-1.326)	(-1.106)	(-0.058)	(-0.104)	(0.023)	(-0.214)	(-0.360)	(-0.287)	(-0.498)	(-0.512)
Liquidity L	0.049	0.030	0.051	-0.007	0.112	0.114	0.114	0.133	0.037	0.037	0.033	0.041
	(0.365)	(0.220)	(0.376)	(-0.054)	(1.280)	(1.278)	(1.378)	(1.349)	(0.651)	(0.672)	(0.575)	(0.698)
Regional econ. act.	-0.519	-0.756	-0.634	-0.148	0.430	0.418	0.525*	$0.564^{*}$	0.208	0.189	0.148	$0.361^{*}$
	(-0.745)	(-1.072)	(-0.875)	(-0.217)	(1.419)	(1.316)	(1.716)	(1.891)	(1.081)	(0.935)	(0.747)	(1.835)
Int. controls x crisis	Yes	$\mathbf{Yes}$	Yes	Yes	Yes	Yes	Yes	$\mathbf{Yes}$	Yes	Yes	Yes	Yes
Int. controls x post-crisis	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Other variables												
Exp. trade finance		-0.131***				-0.015				0.015		
		(-2.742)				(+0.304)				(cuc.u)		
Exp. middle market		0.022 (0.327)				-0.022 (-0.533)				0.438)		
Trade finance x crisis		$0.878^{***}$				0.013				-0.019		

Table 14 – Institutional investors, corporations and individuals – post-crisis

							0.099	(0.649)	-0.353	(-0.679)	Yes		949		8.385	0	42.31	1
			-0.056	(-0.387)	0.331	(0.931)					Yes		949		9.030	0	39.79	1
(-0.190)	-0.239***	(-2.879)									Yes		949		10.56	0	33.94	1
											Yes		949		9.601	0	33.74	1
							0.261	(0.892)	-2.052***	(-3.811)	Yes		949		8.297	0	47.20	1
			0.058	(0.276)	-0.251	(-0.494)					Yes		949		7.581	0	40.67	1
(0.112)	-0.048	(-0.464)									Yes		949		8.973	0	41.61	1
											Yes		949		7.838	0	45.73	1
							-0.677*	(-1.889)	-5.740***	(-5.531)	Yes		949		18.92	0	42.89	1
			-0.00	(-0.042)	0.988	(0.914)					Yes		949		6.160	0	40.16	1
(3.340)	-0.207	(-0.868)									Yes		949		8.329	0	30.77	1
											Yes		949		6.132	1.03e-10	42.70	1
	Middle market x crisis		Foreign funding		Foreign Fund x crisis		Rel. on instit. invest.		Instit. Invest. x crisis		Int. other variables x	post-crisis	Observations	F	F-p	Hansen	hansenp	# of cross sections

# APPENDIX

## **Operational Definitions of Variables Used in the Tests**

1 – Independent Variables

 $\Delta Deposits_{i,t}$  is the change in deposits of bank *i* in period *t*, measured by the first difference of the log of deposits.

The database from the FGC provides the amount of insured and uninsured deposits.

# 2 – Dependent Variables

Our right-hand-side variables are defined as below. Some are used in levels (contemporaneous and/or lagged) and others are used in first differences, as explained in section I.

# 2.1 Risk

The bank-specific risk indicators chosen are commonly used in the literature. The operational definition of all the variables is described below.

Equity: we measure the ratio of equity to total assets to examine capital adequacy.

**Low quality loans:** the assessment of the quality of assets can be made using several indicators. To a great extent, empirical studies use the ratio of nonperforming loans and total assets. We prefer a more forward looking metric: the ratio of low quality loans to total assets. Brazilian banks must rate their credit operations in an ascending order of risk, on levels AA, A, B, C, D, E, F, G and H and report the volume of credit in each of these ratings in their financial statements. Low quality loans are those that fall into one of the ratings E to H. Resolution 2.682 from the Brazilian National Monetary Council states that loans due for more than 90 days should be rated E or worse.

**Liquidity:** We use as a proxy for liquidity (*cash* + *tradable securities*) / *assets*.

# 2.2 Control Variables

**Size:** we measure size as the *natural log of assets*. We include this variable as a fundamental in order to disentangle the pure effect of size on deposits from the special characteristics that may cause a bank to be too big to fail.

**Regional economic activity:** although bank legislation allows banks to open branches and have operations throughout all Brazilian states, many banks focus on specific states to do business. Deposits may thus be influenced by the economic activity of individual states. We use data from the retail sales survey done by the Brazilian Institute for Geography and Statistics (IBGE, for its acronym in Portuguese), which provides the growth in retail sales for each state of the federation as well as the resulting national growth in retail sales<sup>1</sup>. This is the most used indicator of regional economic activity in Brazil. To assign a state to each bank, we use the following procedure: if a bank has branches in more than 10 states<sup>2</sup> and no single state accounts for more than 50% of its

<sup>&</sup>lt;sup>1</sup> The index is released on a monthly basis. We use the 12-month compound growth in retail sales (which does not need to be adjusted for seasonality) for June and December to match with the rest of our data.

<sup>&</sup>lt;sup>2</sup> Brazil has 27 states.

branches, we consider it a nationwide bank, and use the national index. Otherwise we use the index for the state where the bank has more branches. Thus: *Regional Economic Activity*<sub>*i*,*t*</sub> is the change in retail sales index of the state in which the bank has more branches (considers heterogeneous macro-effects over depositors-base).

**Exposure on Loans:** loans are classified into 8 different categories: 1) trade finance (import and export); 2) short-term (less than 12 months) loans to non-financial companies; 3) agricultural; 4) real estate; 5) consumer goods (including auto vehicles); 6) infrastructure; 7) personal (loans made to individuals without specifying a particular purpose); 8) others. Exposure to each of these classes of loans is measured as the ratio between the amount of loans in that class and total assets. For example, exposure to trade finance is measured as the amount of loans qualified as trade finance and total assets. We are especially interested in classes 1 (trade finance) and 2 (working capital). Trade finance loans are very collateralized by import/export contracts, have typically very low delinquency rate and loss given default and are thus expected to be very safe during the crisis. On the other hand, working capital loans have typically low collateral and are mostly issued by small and middle firms (since in Brazil these firms have little access to long term debt markets) and held to maturity by the lender bank (instead of being securitized and traded in the secondary market) and thus we can expect these loans to be riskier during the financial crisis.

**Deposit concentration:** a bank with a larger depositor base is naturally more diversified than another that relies on few depositors to fund its assets. In addition, deposit concentration may create incentives for depositors to "run first" during periods in which informational asymmetry is higher. Deposit concentration is measured as the portion of a bank's asset being funded by each depositor on average, i.e., *total deposits / (total assets \* number of depositors)* in each semester.

**Reliance on institutional investors:** Chen, Goldstein and Jiang (2010) find evidence that the behavior of institutional investors depends on whether they are surrounded by other institutional investors or by retail investors. We use the ratio between the amount of certificates of deposit held by institutional investors and total assets to account for the reliance on institutional investors.

**Reliance on foreign funds:** banks that rely on foreign funds may be more likely to struggle to obtain funding during episodes that reduce the amount of foreign capital to emerging markets in general or to Brazil in particular. We define the reliance on foreign funds as the proportion of assets being funded by sources obtained abroad, i.e. *total foreign funds / total assets*.

**Mechanical Change in deposits:** We also compute for each bank the change occurred in insured and uninsured deposits due to the change in the amount insured in September 2006 (*MechChange*). This computation is based on an unique bank level data on the number of depositors and volume of deposits in several different deposit-size brackets for all Brazilian banking firms. Since the change took place in September 2006, we compute, based on the data of Jun/2006 the amount of uninsured deposits that became insured due to the simple fact that the amount insured was increased. Had we neglected this change, we would end up with a measurement error in our left-hand-side variables (*change in insured deposits* and *change in uninsured deposits*) in the period Dec/2006, since these are not changes derived from depositors moving their resources from one bank to another, which is ultimately what we want to measure. It is also important that, since the change in the amount insured affects each bank differently (because they have different depositor profiles) it is not simply a change in level that could be captured by time fixed effects. As such, in order to correct this measurement error, we create a variable that assumes the value of the change in uninsured deposits due to the increase in the amount insured in Dec/2006 and 0 in all other periods<sup>3</sup>. We create an analogous variable for insured deposits.

# 2.3 Identification variables

We use two different variables that allow us to perform our identification strategy.

**Big Bank:** This is a dummy that assumes 1 for the banks that could be perceived as too big to fail and 0 otherwise. As explained above, these are the eight largest banks up to Jun/2008. In Dec/2008 there are six banks, because there were two mergers between banks belonging to this group in the second semester of 2008. The reasons for choosing these eight banks are described in section I.

**Crisis:** This is a Crisis dummy that assumes 1 for period ending in Dec/2008 and 0 in all other periods.

**Post-Crisis:** This is an indicator of the period that followed the crisis. It assumes 1 for periods ending in Jun/2009 and Dec/2009 and 0 in all other periods.

<sup>&</sup>lt;sup>3</sup> As expected, the coefficient of this variable is equal to 1 in our regressions, with significance of less than 1%.

# **Banco Central do Brasil**

# Trabalhos para Discussão

Os Trabalhos para Discussão podem ser acessados na internet, no formato PDF, no endereço: http://www.bc.gov.br

# **Working Paper Series**

Working Papers in PDF format can be downloaded from: http://www.bc.gov.br

1	<b>Implementing Inflation Targeting in Brazil</b> Joel Bogdanski, Alexandre Antonio Tombini and Sérgio Ribeiro da Costa Werlang	Jul/2000
2	Política Monetária e Supervisão do Sistema Financeiro Nacional no Banco Central do Brasil Eduardo Lundberg	Jul/2000
	Monetary Policy and Banking Supervision Functions on the Central Bank Eduardo Lundberg	Jul/2000
3	<b>Private Sector Participation: a Theoretical Justification of the Brazilian</b> <b>Position</b> <i>Sérgio Ribeiro da Costa Werlang</i>	Jul/2000
4	<b>An Information Theory Approach to the Aggregation of Log-Linear</b> <b>Models</b> <i>Pedro H. Albuquerque</i>	Jul/2000
5	<b>The Pass-Through from Depreciation to Inflation: a Panel Study</b> Ilan Goldfajn and Sérgio Ribeiro da Costa Werlang	Jul/2000
6	<b>Optimal Interest Rate Rules in Inflation Targeting Frameworks</b> José Alvaro Rodrigues Neto, Fabio Araújo and Marta Baltar J. Moreira	Jul/2000
7	<b>Leading Indicators of Inflation for Brazil</b> Marcelle Chauvet	Sep/2000
8	<b>The Correlation Matrix of the Brazilian Central Bank's Standard Model</b> <b>for Interest Rate Market Risk</b> <i>José Alvaro Rodrigues Neto</i>	Sep/2000
9	<b>Estimating Exchange Market Pressure and Intervention Activity</b> <i>Emanuel-Werner Kohlscheen</i>	Nov/2000
10	Análise do Financiamento Externo a uma Pequena Economia Aplicação da Teoria do Prêmio Monetário ao Caso Brasileiro: 1991–1998 Carlos Hamilton Vasconcelos Araújo e Renato Galvão Flôres Júnior	Mar/2001
11	<b>A Note on the Efficient Estimation of Inflation in Brazil</b> <i>Michael F. Bryan and Stephen G. Cecchetti</i>	Mar/2001
12	A Test of Competition in Brazilian Banking Márcio I. Nakane	Mar/2001

13	Modelos de Previsão de Insolvência Bancária no Brasil Marcio Magalhães Janot	Mar/2001
14	<b>Evaluating Core Inflation Measures for Brazil</b> Francisco Marcos Rodrigues Figueiredo	Mar/2001
15	Is It Worth Tracking Dollar/Real Implied Volatility? Sandro Canesso de Andrade and Benjamin Miranda Tabak	Mar/2001
16	Avaliação das Projeções do Modelo Estrutural do Banco Central do Brasil para a Taxa de Variação do IPCA Sergio Afonso Lago Alves	Mar/2001
	<b>Evaluation of the Central Bank of Brazil Structural Model's Inflation Forecasts in an Inflation Targeting Framework</b> <i>Sergio Afonso Lago Alves</i>	Jul/2001
17	<b>Estimando o Produto Potencial Brasileiro: uma Abordagem de Função de Produção</b> <i>Tito Nícias Teixeira da Silva Filho</i>	Abr/2001
	<b>Estimating Brazilian Potential Output: a Production Function Approach</b> <i>Tito Nícias Teixeira da Silva Filho</i>	Aug/2002
18	<b>A Simple Model for Inflation Targeting in Brazil</b> Paulo Springer de Freitas and Marcelo Kfoury Muinhos	Apr/2001
19	<b>Uncovered Interest Parity with Fundamentals: a Brazilian Exchange Rate Forecast Model</b> <i>Marcelo Kfoury Muinhos, Paulo Springer de Freitas and Fabio Araújo</i>	May/2001
20	Credit Channel without the LM Curve Victorio Y. T. Chu and Márcio I. Nakane	May/2001
21	<b>Os Impactos Econômicos da CPMF: Teoria e Evidência</b> <i>Pedro H. Albuquerque</i>	Jun/2001
22	<b>Decentralized Portfolio Management</b> Paulo Coutinho and Benjamin Miranda Tabak	Jun/2001
23	Os Efeitos da CPMF sobre a Intermediação Financeira Sérgio Mikio Koyama e Márcio I. Nakane	Jul/2001
24	Inflation Targeting in Brazil: Shocks, Backward-Looking Prices, and IMF Conditionality Joel Bogdanski, Paulo Springer de Freitas, Ilan Goldfajn and Alexandre Antonio Tombini	Aug/2001
25	Inflation Targeting in Brazil: Reviewing Two Years of Monetary Policy 1999/00 Pedro Fachada	Aug/2001
26	Inflation Targeting in an Open Financially Integrated Emerging Economy: the Case of Brazil Marcelo Kfoury Muinhos	Aug/2001
27	<b>Complementaridade e Fungibilidade dos Fluxos de Capitais Internacionais</b> <i>Carlos Hamilton Vasconcelos Araújo e Renato Galvão Flôres Júnior</i>	Set/2001

28	<b>Regras Monetárias e Dinâmica Macroeconômica no Brasil: uma Abordagem de Expectativas Racionais</b> <i>Marco Antonio Bonomo e Ricardo D. Brito</i>	Nov/2001
29	Using a Money Demand Model to Evaluate Monetary Policies in Brazil Pedro H. Albuquerque and Solange Gouvêa	Nov/2001
30	<b>Testing the Expectations Hypothesis in the Brazilian Term Structure of</b> <b>Interest Rates</b> <i>Benjamin Miranda Tabak and Sandro Canesso de Andrade</i>	Nov/2001
31	Algumas Considerações sobre a Sazonalidade no IPCA Francisco Marcos R. Figueiredo e Roberta Blass Staub	Nov/2001
32	<b>Crises Cambiais e Ataques Especulativos no Brasil</b> <i>Mauro Costa Miranda</i>	Nov/2001
33	Monetary Policy and Inflation in Brazil (1975-2000): a VAR Estimation André Minella	Nov/2001
34	<b>Constrained Discretion and Collective Action Problems: Reflections on the Resolution of International Financial Crises</b> Arminio Fraga and Daniel Luiz Gleizer	Nov/2001
35	<b>Uma Definição Operacional de Estabilidade de Preços</b> <i>Tito Nícias Teixeira da Silva Filho</i>	Dez/2001
36	<b>Can Emerging Markets Float? Should They Inflation Target?</b> <i>Barry Eichengreen</i>	Feb/2002
37	<b>Monetary Policy in Brazil: Remarks on the Inflation Targeting Regime,</b> <b>Public Debt Management and Open Market Operations</b> <i>Luiz Fernando Figueiredo, Pedro Fachada and Sérgio Goldenstein</i>	Mar/2002
38	Volatilidade Implícita e Antecipação de Eventos de <i>Stress</i> : um Teste para o Mercado Brasileiro <i>Frederico Pechir Gomes</i>	Mar/2002
39	<b>Opções sobre Dólar Comercial e Expectativas a Respeito do Comportamento da Taxa de Câmbio</b> <i>Paulo Castor de Castro</i>	Mar/2002
40	<b>Speculative Attacks on Debts, Dollarization and Optimum Currency</b> <b>Areas</b> <i>Aloisio Araujo and Márcia Leon</i>	Apr/2002
41	<b>Mudanças de Regime no Câmbio Brasileiro</b> Carlos Hamilton V. Araújo e Getúlio B. da Silveira Filho	Jun/2002
42	Modelo Estrutural com Setor Externo: Endogenização do Prêmio de Risco e do Câmbio Marcelo Kfoury Muinhos, Sérgio Afonso Lago Alves e Gil Riella	Jun/2002
43	<b>The Effects of the Brazilian ADRs Program on Domestic Market</b> <b>Efficiency</b> <i>Benjamin Miranda Tabak and Eduardo José Araújo Lima</i>	Jun/2002

44	<b>Estrutura Competitiva, Produtividade Industrial e Liberação Comercial no Brasil</b> Pedro Cavalcanti Ferreira e Osmani Teixeira de Carvalho Guillén	Jun/2002
45	<b>Optimal Monetary Policy, Gains from Commitment, and Inflation</b> <b>Persistence</b> <i>André Minella</i>	Aug/2002
46	<b>The Determinants of Bank Interest Spread in Brazil</b> Tarsila Segalla Afanasieff, Priscilla Maria Villa Lhacer and Márcio I. Nakane	Aug/2002
47	Indicadores Derivados de Agregados Monetários Fernando de Aquino Fonseca Neto e José Albuquerque Júnior	Set/2002
48	Should Government Smooth Exchange Rate Risk? Ilan Goldfajn and Marcos Antonio Silveira	Sep/2002
49	<b>Desenvolvimento do Sistema Financeiro e Crescimento Econômico no Brasil: Evidências de Causalidade</b> <i>Orlando Carneiro de Matos</i>	Set/2002
50	Macroeconomic Coordination and Inflation Targeting in a Two-Country Model Eui Jung Chang, Marcelo Kfoury Muinhos and Joanílio Rodolpho Teixeira	Sep/2002
51	<b>Credit Channel with Sovereign Credit Risk: an Empirical Test</b> Victorio Yi Tson Chu	Sep/2002
52	Generalized Hyperbolic Distributions and Brazilian Data José Fajardo and Aquiles Farias	Sep/2002
53	Inflation Targeting in Brazil: Lessons and Challenges André Minella, Paulo Springer de Freitas, Ilan Goldfajn and Marcelo Kfoury Muinhos	Nov/2002
54	Stock Returns and Volatility Benjamin Miranda Tabak and Solange Maria Guerra	Nov/2002
55	<b>Componentes de Curto e Longo Prazo das Taxas de Juros no Brasil</b> Carlos Hamilton Vasconcelos Araújo e Osmani Teixeira de Carvalho de Guillén	Nov/2002
56	Causality and Cointegration in Stock Markets: the Case of Latin America Benjamin Miranda Tabak and Eduardo José Araújo Lima	Dec/2002
57	As Leis de Falência: uma Abordagem Econômica Aloisio Araujo	Dez/2002
58	<b>The Random Walk Hypothesis and the Behavior of Foreign Capital</b> <b>Portfolio Flows: the Brazilian Stock Market Case</b> <i>Benjamin Miranda Tabak</i>	Dec/2002
59	<b>Os Preços Administrados e a Inflação no Brasil</b> Francisco Marcos R. Figueiredo e Thaís Porto Ferreira	Dez/2002
60	<b>Delegated Portfolio Management</b> Paulo Coutinho and Benjamin Miranda Tabak	Dec/2002

61	O Uso de Dados de Alta Freqüência na Estimação da Volatilidade e do Valor em Risco para o Ibovespa João Maurício de Souza Moreira e Eduardo Facó Lemgruber	Dez/2002
62	<b>Taxa de Juros e Concentração Bancária no Brasil</b> Eduardo Kiyoshi Tonooka e Sérgio Mikio Koyama	Fev/2003
63	<b>Optimal Monetary Rules: the Case of Brazil</b> Charles Lima de Almeida, Marco Aurélio Peres, Geraldo da Silva e Souza and Benjamin Miranda Tabak	Feb/2003
64	Medium-Size Macroeconomic Model for the Brazilian Economy Marcelo Kfoury Muinhos and Sergio Afonso Lago Alves	Feb/2003
65	<b>On the Information Content of Oil Future Prices</b> <i>Benjamin Miranda Tabak</i>	Feb/2003
66	<b>A Taxa de Juros de Equilíbrio: uma Abordagem Múltipla</b> Pedro Calhman de Miranda e Marcelo Kfoury Muinhos	Fev/2003
67	Avaliação de Métodos de Cálculo de Exigência de Capital para Risco de Mercado de Carteiras de Ações no Brasil Gustavo S. Araújo, João Maurício S. Moreira e Ricardo S. Maia Clemente	Fev/2003
68	<b>Real Balances in the Utility Function: Evidence for Brazil</b> Leonardo Soriano de Alencar and Márcio I. Nakane	Feb/2003
69	<b>r-filters: a Hodrick-Prescott Filter Generalization</b> Fabio Araújo, Marta Baltar Moreira Areosa and José Alvaro Rodrigues Neto	Feb/2003
70	<b>Monetary Policy Surprises and the Brazilian Term Structure of Interest</b> <b>Rates</b> <i>Benjamin Miranda Tabak</i>	Feb/2003
71	<b>On Shadow-Prices of Banks in Real-Time Gross Settlement Systems</b> <i>Rodrigo Penaloza</i>	Apr/2003
72	<b>O Prêmio pela Maturidade na Estrutura a Termo das Taxas de Juros Brasileiras</b> Ricardo Dias de Oliveira Brito, Angelo J. Mont'Alverne Duarte e Osmani Teixeira de C. Guillen	Maio/2003
73	Análise de Componentes Principais de Dados Funcionais – uma Aplicação às Estruturas a Termo de Taxas de Juros Getúlio Borges da Silveira e Octavio Bessada	Maio/2003
74	Aplicação do Modelo de Black, Derman & Toy à Precificação de Opções Sobre Títulos de Renda Fixa Octavio Manuel Bessada Lion, Carlos Alberto Nunes Cosenza e César das Neves	Maio/2003
75	<b>Brazil's Financial System: Resilience to Shocks, no Currency</b> <b>Substitution, but Struggling to Promote Growth</b> <i>Ilan Goldfajn, Katherine Hennings and Helio Mori</i>	Jun/2003

76	Inflation Targeting in Emerging Market Economies Arminio Fraga, Ilan Goldfajn and André Minella	Jun/2003
77	<b>Inflation Targeting in Brazil: Constructing Credibility under Exchange</b> <b>Rate Volatility</b> <i>André Minella, Paulo Springer de Freitas, Ilan Goldfajn and Marcelo Kfoury</i> <i>Muinhos</i>	Jul/2003
78	<b>Contornando os Pressupostos de Black &amp; Scholes: Aplicação do Modelo de Precificação de Opções de Duan no Mercado Brasileiro</b> <i>Gustavo Silva Araújo, Claudio Henrique da Silveira Barbedo, Antonio</i> <i>Carlos Figueiredo, Eduardo Facó Lemgruber</i>	Out/2003
79	Inclusão do Decaimento Temporal na Metodologia Delta-Gama para o Cálculo do VaR de Carteiras Compradas em Opções no Brasil Claudio Henrique da Silveira Barbedo, Gustavo Silva Araújo, Eduardo Facó Lemgruber	Out/2003
80	Diferenças e Semelhanças entre Países da América Latina: uma Análise de <i>Markov Switching</i> para os Ciclos Econômicos de Brasil e Argentina Arnildo da Silva Correa	Out/2003
81	Bank Competition, Agency Costs and the Performance of the Monetary Policy Leonardo Soriano de Alencar and Márcio I. Nakane	Jan/2004
82	Carteiras de Opções: Avaliação de Metodologias de Exigência de Capital no Mercado Brasileiro Cláudio Henrique da Silveira Barbedo e Gustavo Silva Araújo	Mar/2004
83	<b>Does Inflation Targeting Reduce Inflation? An Analysis for the OECD</b> <b>Industrial Countries</b> <i>Thomas Y. Wu</i>	May/2004
84	<b>Speculative Attacks on Debts and Optimum Currency Area: a Welfare Analysis</b> <i>Aloisio Araujo and Marcia Leon</i>	May/2004
85	<b>Risk Premia for Emerging Markets Bonds: Evidence from Brazilian</b> <b>Government Debt, 1996-2002</b> <i>André Soares Loureiro and Fernando de Holanda Barbosa</i>	May/2004
86	Identificação do Fator Estocástico de Descontos e Algumas Implicações sobre Testes de Modelos de Consumo Fabio Araujo e João Victor Issler	Maio/2004
87	Mercado de Crédito: uma Análise Econométrica dos Volumes de Crédito Total e Habitacional no Brasil Ana Carla Abrão Costa	Dez/2004
88	Ciclos Internacionais de Negócios: uma Análise de Mudança de Regime Markoviano para Brasil, Argentina e Estados Unidos Arnildo da Silva Correa e Ronald Otto Hillbrecht	Dez/2004
89	O Mercado de <i>Hedge</i> Cambial no Brasil: Reação das Instituições Financeiras a Intervenções do Banco Central Fernando N. de Oliveira	Dez/2004

90	Bank Privatization and Productivity: Evidence for Brazil Márcio I. Nakane and Daniela B. Weintraub	Dec/2004
91	<b>Credit Risk Measurement and the Regulation of Bank Capital and</b> <b>Provision Requirements in Brazil – a Corporate Analysis</b> <i>Ricardo Schechtman, Valéria Salomão Garcia, Sergio Mikio Koyama and</i> <i>Guilherme Cronemberger Parente</i>	Dec/2004
92	<b>Steady-State Analysis of an Open Economy General Equilibrium Model</b> <b>for Brazil</b> <i>Mirta Noemi Sataka Bugarin, Roberto de Goes Ellery Jr., Victor Gomes</i> <i>Silva, Marcelo Kfoury Muinhos</i>	Apr/2005
93	Avaliação de Modelos de Cálculo de Exigência de Capital para Risco Cambial Claudio H. da S. Barbedo, Gustavo S. Araújo, João Maurício S. Moreira e Ricardo S. Maia Clemente	Abr/2005
94	Simulação Histórica Filtrada: Incorporação da Volatilidade ao Modelo Histórico de Cálculo de Risco para Ativos Não-Lineares Claudio Henrique da Silveira Barbedo, Gustavo Silva Araújo e Eduardo Facó Lemgruber	Abr/2005
95	<b>Comment on Market Discipline and Monetary Policy by Carl Walsh</b> <i>Maurício S. Bugarin and Fábia A. de Carvalho</i>	Apr/2005
96	O que É Estratégia: uma Abordagem Multiparadigmática para a Disciplina Anthero de Moraes Meirelles	Ago/2005
97	Finance and the Business Cycle: a Kalman Filter Approach with Markov Switching Ryan A. Compton and Jose Ricardo da Costa e Silva	Aug/2005
98	<b>Capital Flows Cycle: Stylized Facts and Empirical Evidences for Emerging Market Economies</b> <i>Helio Mori e Marcelo Kfoury Muinhos</i>	Aug/2005
99	Adequação das Medidas de Valor em Risco na Formulação da Exigência de Capital para Estratégias de Opções no Mercado Brasileiro Gustavo Silva Araújo, Claudio Henrique da Silveira Barbedo,e Eduardo Facó Lemgruber	Set/2005
100	<b>Targets and Inflation Dynamics</b> Sergio A. L. Alves and Waldyr D. Areosa	Oct/2005
101	<b>Comparing Equilibrium Real Interest Rates: Different Approaches to</b> <b>Measure Brazilian Rates</b> <i>Marcelo Kfoury Muinhos and Márcio I. Nakane</i>	Mar/2006
102	Judicial Risk and Credit Market Performance: Micro Evidence from Brazilian Payroll Loans Ana Carla A. Costa and João M. P. de Mello	Apr/2006
103	<b>The Effect of Adverse Supply Shocks on Monetary Policy and Output</b> Maria da Glória D. S. Araújo, Mirta Bugarin, Marcelo Kfoury Muinhos and Jose Ricardo C. Silva	Apr/2006

104	<b>Extração de Informação de Opções Cambiais no Brasil</b> <i>Eui Jung Chang e Benjamin Miranda Tabak</i>	Abr/2006
105	<b>Representing Roommate's Preferences with Symmetric Utilities</b> José Alvaro Rodrigues Neto	Apr/2006
106	<b>Testing Nonlinearities Between Brazilian Exchange Rates and Inflation</b> <b>Volatilities</b> <i>Cristiane R. Albuquerque and Marcelo Portugal</i>	May/2006
107	<b>Demand for Bank Services and Market Power in Brazilian Banking</b> Márcio I. Nakane, Leonardo S. Alencar and Fabio Kanczuk	Jun/2006
108	O Efeito da Consignação em Folha nas Taxas de Juros dos Empréstimos Pessoais Eduardo A. S. Rodrigues, Victorio Chu, Leonardo S. Alencar e Tony Takeda	Jun/2006
109	<b>The Recent Brazilian Disinflation Process and Costs</b> <i>Alexandre A. Tombini and Sergio A. Lago Alves</i>	Jun/2006
110	<b>Fatores de Risco e o Spread Bancário no Brasil</b> Fernando G. Bignotto e Eduardo Augusto de Souza Rodrigues	Jul/2006
111	Avaliação de Modelos de Exigência de Capital para Risco de Mercado do Cupom Cambial Alan Cosme Rodrigues da Silva, João Maurício de Souza Moreira e Myrian Beatriz Eiras das Neves	Jul/2006
112	<b>Interdependence and Contagion: an Analysis of Information</b> <b>Transmission in Latin America's Stock Markets</b> <i>Angelo Marsiglia Fasolo</i>	Jul/2006
113	<b>Investigação da Memória de Longo Prazo da Taxa de Câmbio no Brasil</b> Sergio Rubens Stancato de Souza, Benjamin Miranda Tabak e Daniel O. Cajueiro	Ago/2006
114	<b>The Inequality Channel of Monetary Transmission</b> Marta Areosa and Waldyr Areosa	Aug/2006
115	Myopic Loss Aversion and House-Money Effect Overseas: an Experimental Approach José L. B. Fernandes, Juan Ignacio Peña and Benjamin M. Tabak	Sep/2006
116	<b>Out-Of-The-Money Monte Carlo Simulation Option Pricing: the Join</b> <b>Use of Importance Sampling and Descriptive Sampling</b> <i>Jaqueline Terra Moura Marins, Eduardo Saliby and Joséte Florencio dos</i> <i>Santos</i>	Sep/2006
117	An Analysis of Off-Site Supervision of Banks' Profitability, Risk and Capital Adequacy: a Portfolio Simulation Approach Applied to Brazilian Banks Theodore M. Barnhill, Marcos R. Souto and Benjamin M. Tabak	Sep/2006
118	<b>Contagion, Bankruptcy and Social Welfare Analysis in a Financial</b> <b>Economy with Risk Regulation Constraint</b> <i>Aloísio P. Araújo and José Valentim M. Vicente</i>	Oct/2006

119	A Central de Risco de Crédito no Brasil: uma Análise de Utilidade de Informação Ricardo Schechtman	Out/2006
120	Forecasting Interest Rates: an Application for Brazil Eduardo J. A. Lima, Felipe Luduvice and Benjamin M. Tabak	Oct/2006
121	<b>The Role of Consumer's Risk Aversion on Price Rigidity</b> Sergio A. Lago Alves and Mirta N. S. Bugarin	Nov/2006
122	Nonlinear Mechanisms of the Exchange Rate Pass-Through: a Phillips Curve Model With Threshold for Brazil Arnildo da Silva Correa and André Minella	Nov/2006
123	A Neoclassical Analysis of the Brazilian "Lost-Decades" Flávia Mourão Graminho	Nov/2006
124	<b>The Dynamic Relations between Stock Prices and Exchange Rates:</b> <b>Evidence for Brazil</b> <i>Benjamin M. Tabak</i>	Nov/2006
125	Herding Behavior by Equity Foreign Investors on Emerging Markets Barbara Alemanni and José Renato Haas Ornelas	Dec/2006
126	<b>Risk Premium: Insights over the Threshold</b> José L. B. Fernandes, Augusto Hasman and Juan Ignacio Peña	Dec/2006
127	Uma Investigação Baseada em Reamostragem sobre Requerimentos de Capital para Risco de Crédito no Brasil Ricardo Schechtman	Dec/2006
128	<b>Term Structure Movements Implicit in Option Prices</b> <i>Caio Ibsen R. Almeida and José Valentim M. Vicente</i>	Dec/2006
129	<b>Brazil: Taming Inflation Expectations</b> Afonso S. Bevilaqua, Mário Mesquita and André Minella	Jan/2007
130	<b>The Role of Banks in the Brazilian Interbank Market: Does Bank Type Matter?</b> <i>Daniel O. Cajueiro and Benjamin M. Tabak</i>	Jan/2007
131	Long-Range Dependence in Exchange Rates: the Case of the European Monetary System Sergio Rubens Stancato de Souza, Benjamin M. Tabak and Daniel O. Cajueiro	Mar/2007
132	<b>Credit Risk Monte Carlo Simulation Using Simplified Creditmetrics'</b> <b>Model: the Joint Use of Importance Sampling and Descriptive Sampling</b> <i>Jaqueline Terra Moura Marins and Eduardo Saliby</i>	Mar/2007
133	<b>A New Proposal for Collection and Generation of Information on</b> <b>Financial Institutions' Risk: the Case of Derivatives</b> <i>Gilneu F. A. Vivan and Benjamin M. Tabak</i>	Mar/2007
134	Amostragem Descritiva no Apreçamento de Opções Européias através de Simulação Monte Carlo: o Efeito da Dimensionalidade e da Probabilidade de Exercício no Ganho de Precisão Eduardo Saliby, Sergio Luiz Medeiros Proença de Gouvêa e Jaqueline Terra Moura Marins	Abr/2007

135	<b>Evaluation of Default Risk for the Brazilian Banking Sector</b> <i>Marcelo Y. Takami and Benjamin M. Tabak</i>	May/2007
136	Identifying Volatility Risk Premium from Fixed Income Asian Options Caio Ibsen R. Almeida and José Valentim M. Vicente	May/2007
137	Monetary Policy Design under Competing Models of Inflation Persistence Solange Gouvea e Abhijit Sen Gupta	May/2007
138	<b>Forecasting Exchange Rate Density Using Parametric Models:</b> <b>the Case of Brazil</b> <i>Marcos M. Abe, Eui J. Chang and Benjamin M. Tabak</i>	May/2007
139	Selection of Optimal Lag Length inCointegrated VAR Models with Weak Form of Common Cyclical Features Carlos Enrique Carrasco Gutiérrez, Reinaldo Castro Souza and Osmani Teixeira de Carvalho Guillén	Jun/2007
140	Inflation Targeting, Credibility and Confidence Crises Rafael Santos and Aloísio Araújo	Aug/2007
141	<b>Forecasting Bonds Yields in the Brazilian Fixed income Market</b> Jose Vicente and Benjamin M. Tabak	Aug/2007
142	<b>Crises Análise da Coerência de Medidas de Risco no Mercado Brasileiro de Ações e Desenvolvimento de uma Metodologia Híbrida para o Expected Shortfall</b> Alan Cosme Rodrigues da Silva, Eduardo Facó Lemgruber, José Alberto Rebello Baranowski e Renato da Silva Carvalho	Ago/2007
143	<b>Price Rigidity in Brazil: Evidence from CPI Micro Data</b> Solange Gouvea	Sep/2007
144	<b>The Effect of Bid-Ask Prices on Brazilian Options Implied Volatility: a</b> <b>Case Study of Telemar Call Options</b> <i>Claudio Henrique da Silveira Barbedo and Eduardo Facó Lemgruber</i>	Oct/2007
145	<b>The Stability-Concentration Relationship in the Brazilian Banking</b> <b>System</b> <i>Benjamin Miranda Tabak, Solange Maria Guerra, Eduardo José Araújo</i> <i>Lima and Eui Jung Chang</i>	Oct/2007
146	Movimentos da Estrutura a Termo e Critérios de Minimização do Erro de Previsão em um Modelo Paramétrico Exponencial Caio Almeida, Romeu Gomes, André Leite e José Vicente	Out/2007
147	<b>Explaining Bank Failures in Brazil: Micro, Macro and Contagion Effects</b> (1994-1998) Adriana Soares Sales and Maria Eduarda Tannuri-Pianto	Oct/2007
148	<b>Um Modelo de Fatores Latentes com Variáveis Macroeconômicas para a Curva de Cupom Cambial</b> <i>Felipe Pinheiro, Caio Almeida e José Vicente</i>	Out/2007
149	Joint Validation of Credit Rating PDs under Default Correlation Ricardo Schechtman	Oct/2007

150	A Probabilistic Approach for Assessing the Significance of Contextual Variables in Nonparametric Frontier Models: an Application for Brazilian Banks Roberta Blass Staub and Geraldo da Silva e Souza	Oct/2007
151	<b>Building Confidence Intervals with Block Bootstraps for the Variance</b> <b>Ratio Test of Predictability</b> <i>Eduardo José Araújo Lima and Benjamin Miranda Tabak</i>	Nov/2007
152	<b>Demand for Foreign Exchange Derivatives in Brazil:</b> <b>Hedge or Speculation?</b> <i>Fernando N. de Oliveira and Walter Novaes</i>	Dec/2007
153	Aplicação da Amostragem por Importância à Simulação de Opções Asiáticas Fora do Dinheiro Jaqueline Terra Moura Marins	Dez/2007
154	<b>Identification of Monetary Policy Shocks in the Brazilian Market</b> <b>for Bank Reserves</b> <i>Adriana Soares Sales and Maria Tannuri-Pianto</i>	Dec/2007
155	<b>Does Curvature Enhance Forecasting?</b> <i>Caio Almeida, Romeu Gomes, André Leite and José Vicente</i>	Dec/2007
156	Escolha do Banco e Demanda por Empréstimos: um Modelo de Decisão em Duas Etapas Aplicado para o Brasil Sérgio Mikio Koyama e Márcio I. Nakane	Dez/2007
157	Is the Investment-Uncertainty Link Really Elusive? The Harmful Effects of Inflation Uncertainty in Brazil <i>Tito Nícias Teixeira da Silva Filho</i>	Jan/2008
158	<b>Characterizing the Brazilian Term Structure of Interest Rates</b> Osmani T. Guillen and Benjamin M. Tabak	Feb/2008
159	<b>Behavior and Effects of Equity Foreign Investors on Emerging Markets</b> <i>Barbara Alemanni and José Renato Haas Ornelas</i>	Feb/2008
160	<b>The Incidence of Reserve Requirements in Brazil: Do Bank Stockholders</b> <b>Share the Burden?</b> <i>Fábia A. de Carvalho and Cyntia F. Azevedo</i>	Feb/2008
161	<b>Evaluating Value-at-Risk Models via Quantile Regressions</b> Wagner P. Gaglianone, Luiz Renato Lima and Oliver Linton	Feb/2008
162	<b>Balance Sheet Effects in Currency Crises: Evidence from Brazil</b> Marcio M. Janot, Márcio G. P. Garcia and Walter Novaes	Apr/2008
163	Searching for the Natural Rate of Unemployment in a Large Relative Price Shocks' Economy: the Brazilian Case Tito Nícias Teixeira da Silva Filho	May/2008
164	<b>Foreign Banks' Entry and Departure: the recent Brazilian experience</b> (1996-2006) <i>Pedro Fachada</i>	Jun/2008
165	Avaliação de Opções de Troca e Opções de <i>Spread</i> Européias e Americanas	Jul/2008
	Giuliano Carrozza Uzêda Iorio de Souza, Carlos Patrício Samanez e Gustavo Santos Raposo	

166	<b>Testing Hyperinflation Theories Using the Inflation Tax Curve: a case study</b> <i>Fernando de Holanda Barbosa and Tito Nícias Teixeira da Silva Filho</i>	Jul/2008
167	O Poder Discriminante das Operações de Crédito das Instituições Financeiras Brasileiras Clodoaldo Aparecido Annibal	Jul/2008
168	An Integrated Model for Liquidity Management and Short-Term Asset Allocation in Commercial Banks Wenersamy Ramos de Alcântara	Jul/2008
169	Mensuração do Risco Sistêmico no Setor Bancário com Variáveis Contábeis e Econômicas Lucio Rodrigues Capelletto, Eliseu Martins e Luiz João Corrar	Jul/2008
170	Política de Fechamento de Bancos com Regulador Não-Benevolente: Resumo e Aplicação Adriana Soares Sales	Jul/2008
171	Modelos para a Utilização das Operações de Redesconto pelos Bancos com Carteira Comercial no Brasil Sérgio Mikio Koyama e Márcio Issao Nakane	Ago/2008
172	<b>Combining Hodrick-Prescott Filtering with a Production Function</b> <b>Approach to Estimate Output Gap</b> <i>Marta Areosa</i>	Aug/2008
173	<b>Exchange Rate Dynamics and the Relationship between the Random</b> <b>Walk Hypothesis and Official Interventions</b> <i>Eduardo José Araújo Lima and Benjamin Miranda Tabak</i>	Aug/2008
174	Foreign Exchange Market Volatility Information: an investigation of real-dollar exchange rate Frederico Pechir Gomes, Marcelo Yoshio Takami and Vinicius Ratton Brandi	Aug/2008
175	<b>Evaluating Asset Pricing Models in a Fama-French Framework</b> Carlos Enrique Carrasco Gutierrez and Wagner Piazza Gaglianone	Dec/2008
176	Fiat Money and the Value of Binding Portfolio Constraints Mário R. Páscoa, Myrian Petrassi and Juan Pablo Torres-Martínez	Dec/2008
177	<b>Preference for Flexibility and Bayesian Updating</b> <i>Gil Riella</i>	Dec/2008
178	An Econometric Contribution to the Intertemporal Approach of the Current Account Wagner Piazza Gaglianone and João Victor Issler	Dec/2008
179	Are Interest Rate Options Important for the Assessment of Interest Rate Risk? Caio Almeida and José Vicente	Dec/2008
180	A Class of Incomplete and Ambiguity Averse Preferences Leandro Nascimento and Gil Riella	Dec/2008
181	Monetary Channels in Brazil through the Lens of a Semi-Structural Model André Minella and Nelson F. Souza-Sobrinho	Apr/2009

182	Avaliação de Opções Americanas com Barreiras Monitoradas de Forma Discreta Giuliano Carrozza Uzêda Iorio de Souza e Carlos Patrício Samanez	Abr/2009
183	Ganhos da Globalização do Capital Acionário em Crises Cambiais Marcio Janot e Walter Novaes	Abr/2009
184	<b>Behavior Finance and Estimation Risk in Stochastic Portfolio</b> <b>Optimization</b> <i>José Luiz Barros Fernandes, Juan Ignacio Peña and Benjamin</i> <i>Miranda Tabak</i>	Apr/2009
185	Market Forecasts in Brazil: performance and determinants Fabia A. de Carvalho and André Minella	Apr/2009
186	<b>Previsão da Curva de Juros: um modelo estatístico com variáveis macroeconômicas</b> André Luís Leite, Romeu Braz Pereira Gomes Filho e José Valentim Machado Vicente	Maio/2009
187	The Influence of Collateral on Capital Requirements in the Brazilian Financial System: an approach through historical average and logistic regression on probability of default Alan Cosme Rodrigues da Silva, Antônio Carlos Magalhães da Silva, Jaqueline Terra Moura Marins, Myrian Beatriz Eiras da Neves and Giovani Antonio Silva Brito	Jun/2009
188	<b>Pricing Asian Interest Rate Options with a Three-Factor HJM Model</b> Claudio Henrique da Silveira Barbedo, José Valentim Machado Vicente and Octávio Manuel Bessada Lion	Jun/2009
189	Linking Financial and Macroeconomic Factors to Credit Risk Indicators of Brazilian Banks Marcos Souto, Benjamin M. Tabak and Francisco Vazquez	Jul/2009
190	<b>Concentração Bancária, Lucratividade e Risco Sistêmico: uma abordagem de contágio indireto</b> <i>Bruno Silva Martins e Leonardo S. Alencar</i>	Set/2009
191	<b>Concentração e Inadimplência nas Carteiras de Empréstimos dos Bancos Brasileiros</b> <i>Patricia L. Tecles, Benjamin M. Tabak e Roberta B. Staub</i>	Set/2009
192	Inadimplência do Setor Bancário Brasileiro: uma avaliação de suas medidas Clodoaldo Aparecido Annibal	Set/2009
193	<i>Loss Given Default</i> : um estudo sobre perdas em operações prefixadas no mercado brasileiro Antonio Carlos Magalhães da Silva, Jaqueline Terra Moura Marins e Myrian Beatriz Eiras das Neves	Set/2009
194	<b>Testes de Contágio entre Sistemas Bancários – A crise do subprime</b> Benjamin M. Tabak e Manuela M. de Souza	Set/2009
195	From Default Rates to Default Matrices: a complete measurement of Brazilian banks' consumer credit delinquency Ricardo Schechtman	Oct/2009

196	<b>The role of macroeconomic variables in sovereign risk</b> <i>Marco S. Matsumura and José Valentim Vicente</i>	Oct/2009
197	Forecasting the Yield Curve for Brazil Daniel O. Cajueiro, Jose A. Divino and Benjamin M. Tabak	Nov/2009
198	Impacto dos Swaps Cambiais na Curva de Cupom Cambial: uma análise segundo a regressão de componentes principais Alessandra Pasqualina Viola, Margarida Sarmiento Gutierrez, Octávio Bessada Lion e Cláudio Henrique Barbedo	Nov/2009
199	<b>Delegated Portfolio Management and Risk Taking Behavior</b> José Luiz Barros Fernandes, Juan Ignacio Peña and Benjamin Miranda Tabak	Dec/2009
200	<b>Evolution of Bank Efficiency in Brazil: A DEA Approach</b> <i>Roberta B. Staub, Geraldo Souza and Benjamin M. Tabak</i>	Dec/2009
201	Efeitos da Globalização na Inflação Brasileira Rafael Santos e Márcia S. Leon	Jan/2010
202	<b>Considerações sobre a Atuação do Banco Central na Crise de 2008</b> Mário Mesquita e Mario Torós	Mar/2010
203	Hiato do Produto e PIB no Brasil: uma Análise de Dados em Tempo Real Rafael Tiecher Cusinato, André Minella e Sabino da Silva Pôrto Júnior	Abr/2010
204	Fiscal and monetary policy interaction: a simulation based analysis of a two-country New Keynesian DSGE model with heterogeneous households Marcos Valli and Fabia A. de Carvalho	Apr/2010
205	Model selection, estimation and forecasting in VAR models with short-run and long-run restrictions George Athanasopoulos, Osmani Teixeira de Carvalho Guillén, João Victor Issler and Farshid Vahid	Apr/2010
206	<b>Fluctuation Dynamics in US interest rates and the role of monetary policy</b> <i>Daniel Oliveira Cajueiro and Benjamin M. Tabak</i>	Apr/2010
207	<b>Brazilian Strategy for Managing the Risk of Foreign Exchange Rate</b> <b>Exposure During a Crisis</b> <i>Antonio Francisco A. Silva Jr.</i>	Apr/2010
208	<b>Correlação de <i>default</i>: uma investigação empírica de créditos de varejo no Brasil</b> Antonio Carlos Magalhães da Silva, Arnildo da Silva Correa, Jaqueline Terra Moura Marins e Myrian Beatriz Eiras das Neves	Maio/2010
209	<b>Produção Industrial no Brasil: uma análise de dados em tempo real</b> Rafael Tiecher Cusinato, André Minella e Sabino da Silva Pôrto Júnior	Maio/2010
210	<b>Determinants of Bank Efficiency: the case of Brazil</b> Patricia Tecles and Benjamin M. Tabak	May/2010

211	<b>Pessimistic Foreign Investors and Turmoil in Emerging Markets: the</b> <b>case of Brazil in 2002</b> <i>Sandro C. Andrade and Emanuel Kohlscheen</i>	Aug/2010
212	<b>The Natural Rate of Unemployment in Brazil, Chile, Colombia and Venezuela: some results and challenges</b> <i>Tito Nícias Teixeira da Silva</i>	Sep/2010
213	<b>Estimation of Economic Capital Concerning Operational Risk in a Brazilian banking industry case</b> Helder Ferreira de Mendonça, Délio José Cordeiro Galvão and Renato Falci Villela Loures	Oct/2010
214	<b>Do Inflation-linked Bonds Contain Information about Future Inflation?</b> José Valentim Machado Vicente and Osmani Teixeira de Carvalho Guillen	Oct/2010
215	<b>The Effects of Loan Portfolio Concentration on Brazilian Banks' Return</b> <b>and Risk</b> <i>Benjamin M. Tabak, Dimas M. Fazio and Daniel O. Cajueiro</i>	Oct/2010
216	<b>Cyclical Effects of Bank Capital Buffers with Imperfect Credit Markets:</b> <b>international evidence</b> <i>A.R. Fonseca, F. González and L. Pereira da Silva</i>	Oct/2010
217	<b>Financial Stability and Monetary Policy – The case of Brazil</b> Benjamin M. Tabak, Marcela T. Laiz and Daniel O. Cajueiro	Oct/2010
218	<b>The Role of Interest Rates in the Brazilian Business Cycles</b> <i>Nelson F. Souza-Sobrinho</i>	Oct/2010
219	<b>The Brazilian Interbank Network Structure and Systemic Risk</b> Edson Bastos e Santos and Rama Cont	Oct/2010
220	Eficiência Bancária e Inadimplência: testes de Causalidade Benjamin M. Tabak, Giovana L. Craveiro e Daniel O. Cajueiro	Out/2010
221	<b>Financial Instability and Credit Constraint: evidence from the cost of bank financing</b> <i>Bruno S. Martins</i>	Nov/2010
222	<b>O Comportamento Cíclico do Capital dos Bancos Brasileiros</b> <i>R. A. Ferreira, A. C. Noronha, B. M. Tabak e D. O. Cajueiro</i>	Nov/2010
223	<b>Forecasting the Yield Curve with Linear Factor Models</b> Marco Shinobu Matsumura, Ajax Reynaldo Bello Moreira and José Valentim Machado Vicente	Nov/2010
224	<b>Emerging Floaters: pass-throughs and (some) new commodity</b> <b>currencies</b> <i>Emanuel Kohlscheen</i>	Nov/2010
225	<b>Expectativas Inflacionárias e Inflação Implícita no Mercado Brasileiro</b> <i>Flávio de Freitas Val, Claudio Henrique da Silveira Barbedo e</i> <i>Marcelo Verdini Maia</i>	Nov/2010
226	A Macro Stress Test Model of Credit Risk for the Brazilian Banking Sector Francisco Vazquez, Benjamin M.Tabak and Marcos Souto	Nov/2010

227	Uma Nota sobre Erros de Previsão da Inflação de Curto Prazo Emanuel Kohlscheen	Nov/2010
228	<b>Forecasting Brazilian Inflation Using a Large Data Set</b> <i>Francisco Marcos Rodrigues Figueiredo</i>	Dec/2010
229	<b>Financial Fragility in a General Equilibrium Model: the Brazilian case</b> <i>Benjamin M. Tabak, Daniel O. Cajueiro and Dimas M. Fazio</i>	Dec/2010
230	<b>Is Inflation Persistence Over?</b> <i>Fernando N. de Oliveira and Myrian Petrassi</i>	Dec/2010
231	<b>Capital Requirements and Business Cycles with Credit Market</b> <b>Imperfections</b> <i>P. R. Agénor, K. Alper and L. Pereira da Silva</i>	Jan/2011
232	Modeling Default Probabilities: the case of Brazil Benjamin M. Tabak, Daniel O. Cajueiro and A. Luduvice	Jan/2011